## **DELTA MOSQUITO & VECTOR CONTROL DISTRICT**

Conlin Reis General Manager

Paul Harlien Operations Program Manager

Mary Ellen Gomez Administrative Assistant

Andrea Troupin Scientific Program Manager 1737 West Houston Avenue \* Visalia, California 93291 Phone (559) 732-8606 \* (877) 732-8606 \* Fax (559)-732-7441 www.deltamvcd.gov



Rick Alvarez Vector Control Supervisor

> Bryan Ruiz Supervisor Assistant

Bryan Ferguson *Foreman* 

Erick Arriaga Community Education & Outreach Coordinator

- DATE: Friday, July 5, 2024
- TO: Board of Trustees, Delta Mosquito and Vector Control District (DMVCD)
- FROM: Conlin Reis, General Manager
- SUBJECT: Regular Meeting of the District's Board of Trustees
- TIME: Wednesday, July 10, 2024, at 4:30 p.m.
- PLACE: District Boardroom, 1737 West Houston Avenue, Visalia CA, 93291

### AGENDA:

- 1. Roll Call
- 2. Introduction of Conlin Reis, General Manager

### 3. Employee of the Quarter

The General Manager and the Board of Trustees will recognize and present the Employee of the Quarter Certificate to Theodore McGill.

### 4. Public Forum (Limited to three minutes per speaker)

- a) Members of the public may comment on any item not on the agenda that is within the jurisdiction of the Board of Trustees (Board). Under state law, matters presented during public comment <u>cannot be discussed or acted upon</u> by the Board in this meeting.
- b) For items on the agenda, the public is invited to make comments during the public comment period.
- c) Any person addressing the Board will be limited to a maximum of three (3) minutes. Public comments will be limited to a total of 15 minutes during the public comment period.

- d) If there are more than five (5) people wishing to comment, then time will be divided equally between all people wishing to speak, so that everyone has an opportunity to address the Board.
- e) Public comments may be submitted via email to publiccomments@deltamvcd.gov

### ACTION 5. Consent Calendar

- f) June Minutes
- g) June Bills (Board Order #'s 43-45)
- h) July Payroll & Bills (Board Order #1)

### 6. Staff Presentations

The Operations and Laboratory Staff will give a brief summary on events in their Department.

### 7. Manager's Report

The General Manager will report on items of Delta Mosquito and Vector Control District (DMVCD) operational and laboratory interest.

### 8. Adoption of CalPERS 457 Program

The Board will consider adoption of the 457 deferred compensation program offered by CalPERS for optional use by employees. This includes an optional loan provision.

- A. Approval of Employer Adoption Agreement
- B. Resolution #2025-01 Adopting CalPERS 457 Plan
- C. Authorization for Admin to Complete Enrollment.

### 9. Rescind or Modify Policy 2139, District Equipment Loan Policy

The Board will consider modification or rescission of Policy 2139, District Equipment Loan Policy to comport with state law regarding use of public resources.

### 10. Manager's District Review and First Steps Presentation

The Manager will give a brief presentation on his initial plan for reviewing District operations and will receive input from the Board.

### **11. Board of Trustees Member Comments**

The Board of Trustees members will have a chance to make any additional comments regarding items within the jurisdiction of the District.

### 12. Future Agenda Items

The Board of Trustees members will have a chance to add to the future Agenda items if they choose to.

### ACTION 13. Adjournment

Adjourn Meeting of the Board of Trustees to reconvene on Wednesday, August 14, 2024, at 4:30 p.m.in the Delta Mosquito and Vector Control District Boardroom, 1737 W. Houston Ave., Visalia, CA.

**<u>Note</u>**: Items designated for information are appropriate for Board action if the Board wishes to act.

### 1. Roll Call

### 2. Introduction of Conlin Reis, General Manager

### **3. Employee of the Quarter**

### 4. Public Forum

### 5. Consent Calendar

## **DELTA MOSQUITO & VECTOR CONTROL DISTRICT**

Paul Harlien Interim Manager Operations Program Manager

Mary Ellen Gomez Administrative Assistant

Andrea Troupin Scientific Program Manager

Erick Arriaga Community Education & Outreach Coordinator 1737 West Houston Avenue, Visalia, California, 93291 Phone: (559) 732-8606 Toll Free: (877) 732-8606 Fax: (559) 732-7441

Rick Alvarez Vector Control Supervisor

> Bryan Ruiz Supervisor Assistant

Bryan Ferguson Foreman

Minutes of the Board of Trustees – Wednesday, June 12, 2024, Start: 4:31 p.m.

### 1. Roll Call:

Present: Greg Gomez, President; Linda Guttierrez, Secretary, Rosemary Hellwig, Larry Roberts, Kevin Caskey, Lori Berger, and Mike Burchett.

Absent: None

Staff: Paul Harlien, Interim Manager/Operations Program Manager; Mary Ellen Gomez, Administrative Assistant

### 2. Public Forum:

None.

### 3. Consent Calendar:

Following discussion, it was moved by Larry Roberts, seconded by Kevin Caskey and the members of the Board of Trustees unanimously approved the consent calendar as presented.

### 4. Manager's Report:

Paul Harlien, Interim Manager provided an update on Delta Mosquito and Vector Control District (DMVCD) May's 2024 Operations. Notified the Board of Trustees that there are some fleet vehicles that have essentially met their lifespan as replacements parts are getting harder to get, considering purchasing Ford Maverick's to replace the aging Chevy Colorados.

Drone treatments have been performed by Bryan Ferguson at the Woodlake rock plant and the Ivanhoe sewer farm, drone treatment is the best option due to heavy vegetation.



The DMVCD Community Education and Outreach Coordinator participated in three outreach events: Lion's Breakfast Club Car Show, Senior Day at Mooney Grove Park and 1<sup>st</sup> of 6 events at the Dinuba Farmers Market.

Andrea Troupin discussed routine mosquito trapping that continued through May 2024. Traps were set throughout the District in Goshen, Traver, London, Delft Colony, Dinuba, Orosi, Seville, Kings River area, Elderwood, Woodlake, Lemon Cove, Ivanhoe, Exeter, Farmersville, and Visalia.

As of June 3, 2024, DMVCD tested 910 of the 1,008 mosquito samples collected in May. The dead bird brain samples collected for this year have all tested negative for WNV, SLEV, and WEEV. No samples were positive for a mosquito-borne disease.

There were 48 technician and homeowner mosquitofish requests in May 2024 to distribute 2,286 fish. In May 2023, there were 53 technician and homeowner mosquitofish requests to distribute 2,974 fish. Approximately 2,113 mosquitofish fry were produced by May 31, 2024.

### 5. Public Hearing:

The Board President opened the floor up for Arcelia Herrera and Emanuel Peterson from SCI to discuss the continuation of the District's Prop 218, i.e., Benefit Assessment. Since no members of the Public were present, the Board of Trustees President verified that no protest letters were received by the DMVCD, and then closed the Public Hearing.

### 6. <u>Resolution 2024-04:</u>

Emanuel Peterson discussed Resolution 2024-04, i.e., the Engineer's Report and levied the annual assessment. Following the discussion, it was moved by Lori Berger, seconded by Linda Guttierrez and the Resolution was unanimously approved.

### 7. Purchase of Ford Maverick:

Following discussion, it was moved by Larry Roberts, seconded by Mike Burchett and the members of the Board of Trustees approved the outright purchase of the Ford Maverick.

### 8. 2024-2025 Fiscal Budget and Resolution 2024-05:

Mary Ellen Gomez, Administrative Analyst presented the Draft 2024-2025 Fiscal Budget and Resolution 2024-05 to the Board of Trustees. Following discussion, it was moved by Mike Burchett, seconded by Rosemary Hellwig to approve the documents presented.

### 9. Board of Trustees Member Comments:

Mike Burchett briefly discussed a resident he knows that was extremely happy with services provided on a pool inspection a technician performed.

### 10. Future Agenda Items:

Greg Gomez mentioned that he would like to see salaries reviewed across the board. Lori Berger mentioned reports again for different divisions during the season. Lastly, initiated by Rosemary Hellwig the Board would like to take the new General Manager to lunch once he starts with the district and have Paul Harlien in attendance.

### 11. Adjournment:

Greg Gomez, Board President adjourned the meeting of the DMVCD's Board of Trustees at 5:29pm.

Paul Harlien, Recording Secretary

CLAIM #	PAYEE	DESCRIPTION		Budget Line Item	AMOUNT
39007	AUTOZONE				156.92
		10 pack bulbs for Chevy Colorados	19.52	Vehicle Supplies	
		(2) exterior door handles and (1) interior door handle (T6)	158.38	Vehicle Supplies	
		Radiator T15	146.35	Vehicle Supplies	
		Warranty Credit Radiator T15	-146.35	Vehicle Supplies	
		Deep Cycle Battery (Guardian ULV)	120.15	Vehicle Supplies	
		Misc Credits Issued	-141.13	Vehicle Supplies	
39008	AWARDS AND SIGNS	Desk Name Plate - Conlin Reis		Office Supplies	56.96
39009	CENTRAL VALLEY BUSINESS FORMS	1000 regular envelopes and 2500 security window envelopes		Office Supplies	504.14
39010	CITY OF VISALIA	Utilities		Utilities	120.35
39011	COMCAST	Internet		Telephone/Cellphones	350.61
39012	UNIFIRST				1,486.62
		Uniforms	1254.86	Uniforms	
		Janitorial	231.76	Janitorial	
39013	VERIZON	Cellphones		Telephone/Cellphones	1,163.08
		TOTAL BOARD ORDER #43			\$3,838.68

CLAIM #	PAYEE	DESCRIPTION	Budget Line Item	AMOUNT
39014	ZACKARY BARRAGAN	Laboratory Tech I		1.368.58
	DANNY CADENA	Vector Control Tech		1,188.83
39016	JOSE CARRERA	Laboratory Tech I		1.099.82
39017	CATHERINE CONROY	Laboratory Tech I		1,287.21
39018	ALYSIA DAVIS	Vector Control Tech I		1,600.64
39019	ARTURO GARCIA	Vector Control Tech I		1,164.35
39020	JOEY GASCA	Vector Control Tech I		1,072.88
39021	JAKE MALDONADO	Laboratory Tech I		1,505.67
39022	MARCO MARTINEZ	Vector Control Tech II		1,864.33
39023	JOSHUA MITCHELL	Laboratory Tech I		1,179.70
39024	ISAIAH ORTANEZ	Laboratory Tech I		886.16
39025	TIMOTHY PARK	Laboratory Tech I		1,179.70
39026	RAFAEL RAMIREZ	Vector Control Tech I		1,189.66
39027	BRENDA RODRIGUEZ-VIVAR	Laboratory Tech II		1,685.98
39028	LISA SALGADO	Vector Control Tech I		1,481.86
39029	ADRIAN SIFUENTES	Vector Control Tech I		1,487.16
39030	RYAN SPRATT	Vector Control Tech I		1,394.52
39031	ISAEL VENEGAS-ALAMO	Vector Control Tech I		1,313.49
39032	KORY WILSON	Vector Control Tech I		1,462.84
39033	DELTA VECTOR CONTROL DIST - EFTPS	Social Security/ Medicare/ Federal Income Tax		6,417.62
39034	DELTA VECTOR CONTROL DIST - EFTPS	State Income Tax		549.89
	DELTA VECTOR CONTROL DIST - EFTPS	CalPERS Retirement		1,630.97
		Total Seasonal Payroll		\$34,011.86
39036	FED EX	Shipping of box to DMVCD from M. Debboun- District laptop and cellphone	Misc. Expense	43.86
	MONARCH FORD	Purchase of 2024 Ford Maverick	Vehicle - Capital	30.470.33
	VALLEY PACIFIC	Fuel	Fuel	4,221.55
	VISALIA POLICE DEPT.	Alarm Permit Renewal	Professional Services	17.40
		TOTAL BILLS		\$34,753.14
		TOTAL BOARD ORDER #44		\$68,765.00

	PAYEE	DESCRIPTION		Budget Line Item	AMOUNT
39040	CA WATER SERVICE	Utilities		Utilities	238.32
39041	CLINE'S	Monthly Service Contract		Maint. Contract	118.09
39042	LOZANO SMITH	Draft Termination Letter - Mustapha Debboun and Guidance re: Consulting		Professional Services	135
39043	SO CAL GAS	Utilities		Utilities	31.69
39044	US BANK				9947.02
		Amazon - Brother Printer (Paul Harlien Office)	433.99	Office Supplies	
		Amazon - Credit Return of Damaged Brother Printer		Office Supplies	
		USPS - Cert and Regular Mailpiece US Treasury	9.58	Office Supplies	
		Ebay - Rear Bumper Chevy Colorado T22		Vehicle Supplies	
		Best Buy - Keyboards, mouse and SD Card - lab computers		Office Supplies	
		Amazon - (14) Igloo 2gal water jugs		Safety Supplies	
		Amazon - 5 tab binder dividers		Office Supplies	
		Amazon - (3) 3 pack ice removal shovels		Fish Supplies	
		Amazon - Brochure Holders - Front Desk		Office Supplies	
		Amazon - Nitrile Gloves Large and X-Large		Safety Supplies	
		Amazon - Tire Plugs		Vehicle Supplies	
		Smart n Final - Multifold paper towels		Janitorial	
		Costco - 3 pack Lysol Wipes, Toilet Paper and Clorox Wipes		Janitorial	
		Best Buy - (3) Laptops for Lab		Office Supplies	
		Verizon Wireless - Cellphones		Telephones/Cellphones	
		Google Suites - Monthly Contract		Maint. Contract	
		SipTrunk - Efax		Telephones/Cellphones	
		Amazon - Toner Mark Nakata		Office Supplies	
		Office Depot - (2) 3000Ct Address labels		Office Supplies	
		Zira - TimeKeeping System		Subscriptions	
		Amazon - (5) SOLO Backpack sprayers		Spray Equipment	
		Amazon - Vehicle Keytags		Vehicle Supplies	
		Amazon - Report Covers		Office Supplies	
		Amazon - (15) All Weather Mats Chevy Colorados		Vehicle Supplies	
		USPS - Cert Mailpiece to Dr. Debboun and (2) rolls stamps		Office Supplies	
		Ford Motor Co All Weather Floor Mats - Outreach Van		Vehicle Supplies	
		FilterBuy - Lab AC Filters	70.96	Building Maintenance	
		Harbor Freight - (1) hook and loop, (1) outlet	30.36	Lab Supplies	
		Best Buy - (1) Micro SD Card and (3) Mouse pads	48.77	Lab Supplies	
		County of Tulare Surplus - (2) monitors/general office supplies	45.40	Lab Supplies	
		WalMart - (3) HDMI cables	33.75	Lab Supplies	
		Lowes - Dawn Soap	12.94	Janitorial	
		Lowes - (2) stone pavers, (4) grip rods, (1) power strip, bleach	52.90	Lab Supplies	
		Key Evidence - (6) key copies sewer farms and (3) key copies Dinuba	56.10	Building Maintenance	
		US Chef Store - (10) 50lb bags of sugar	355.90	Lab Supplies	
		US Chef Store - (5) 12pack cases of 2lb bags of yeast	740.45	Lab Supplies	
		The Pond Guys - (3) MinnFinn Treatments		Fish Supplies	
		Lowes - (1) 100' hose		Yard Maintenance	
		Sportsmans Warehouse - (4) minnow trap clips		Fish Supplies	
		Sportsmans Warehouse - (1) minnow trap clip and (1) 3/4oz weight		Fish Supplies	
		Home Depot - (6) 40lb bags of salt		Fish Supplies	
		Home Depot - Trash bags, (2) spray bottles and (2) car dash cleaners		Janitorial	
		Home Depot - (2) hose nozzles, (2) 2 pack doorstops and (1) deck box		Building Maintenance	
	<u> </u>	Home Depot - (2) hose hozzles, (2) 2 pack doorstops and (1) deck box		Building Maintenance	
		Tractor Supply - (1) pair rubber boots and (2) pairs of work boots		Lab Supplies	
		DreamHost - Monthly Contract		Maint, Contract	
	<u> </u>	Martinizing Green Cleaning - laundering of Outreach Tablecloths		Mint. Contract Misc. Expense	
		Costco - Birthday Treats		Misc. Expense	
		Harbor Freight - (4) outlet strips, 15'extension cord and black zip ties		Office Supplies	
		B&H Photo - Camera Lens		Public Relations	
		Apple.Com - Additional Phone Storage Erick Arriaga		Subscriptions	
		ZOHO Social Media Archiving		Public Relations	
		Lowes - (2) Portable cooling units		Building Maintenance	
		Andy's Automotive Paint - White base and clear coat T22	63.90	Vehicle Supplies	
	<u> </u>	Total Board Order # 45			\$10,470.12

VOUCHER	PAYEE	DESCRIPTION		Budget Line Item	AMOUNT
39045		GENERAL MANAGER			11,519.98
39046	RICK ALVAREZ	VECTOR CONTROL SUPERVISOR			6,763.73
39047	ERICK ARRIAGA	COMMUNITY EDUCATION & OUTREACH COORDINATOR			5,354.88
39048	BRYAN FERGUSON	FOREMAN			6,694.92
39049	ALEJANDRA GILL	LABORATORY TECH III			4,281.34
39050	MARY ELLEN GOMEZ	ADMINISTRATIVE ASSISTANT			5,906.25
39051	PAUL HARLIEN	OPERATIONS PROGRAM MANAGER			7,339.94
39052	REBECCA HARLIEN	ADMINSTRATIVE ANALYST			5,862.78
39053	LANDON MCGILL	VECTOR CONTROL TECHNICIAN III/MECHANIC			4,437.25
39054	THEODORE MCGILL	VECTOR CONTROL TECHNICIAN III			4,039.53
39055	MARK NAKATA	BIOLOGIST			6,257.15
39056	JUAN PABLO ORTEGA	BIOLOGIST			5,372.30
39057	CARLOS RODRIGUEZ	VECTOR CONTROL TECHNIICIAN III			4,533.91
39058	BRYAN RUIZ				5,283.80
39059	MARIO SANCHEZ				5,360.20
39060	ANDREA TROUPIN	SCIENTIFIC PROGRAM MANAGER			6,481.03
39061	GREG GOMEZ	TRUSTEE PAYROLL - QUARTERLY			277.05
39062	LINDA GUTTIERREZ	TRUSTEE PAYROLL - QUARTERLY			277.05
39063	ROSEMARY HELLWIG	TRUSTEE PAYROLL - QUARTERLY			277.05
39064	KEVIN CASKEY	TRUSTEE PAYROLL - QUARTERLY			277.05
39065	LARRY ROBERTS	TRUSTEE PAYROLL - QUARTERLY			277.05
39066	LORI BERGER	TRUSTEE PAYROLL - QUARTERLY			277.05
39067	MICHAEL BURCHETT	TRUSTEE PAYROLL - QUARTERLY			277.05
		Sub-Total Full-Time and Trustee Payroll			\$97,428.34
					<b>401,420.04</b>
39068	VSP	Vision Plan Premium			576.53
39069	DELTA DENTAL PLAN	Dental Plan Premium			1,496.14
39070	LINCOLN FINANCIAL GROUP	Life/STD & LTD Insurance			1,501.47
39071	DELTA VECTOR CONTROL DIST - EFTPS	CalPERS Health Insurance Premium			30,743.50
39072	DELTA VECTOR CONTROL DIST - EFTPS	Social Security/ Medicare/ Federal Income Tax		Employee 71% - District 29%	34,711.04
39073	DELTA VECTOR CONTROL DIST - EFTPS	State Income Tax	ĺ	Employee 100%	6,167.50
39074	DELTA VECTOR CONTROL DIST - EFTPS	CalPERS Retirement		Employee 40% - District 60%	25,698.07
39075	DELTA VECTOR CONTROL DIST - EFTPS	Quarterly UI Payment			7,727.23
39076	DELTA VECT CONT DIST	Flex Benefit Plan		Employee 100%	1,083.29
39077	MISSION SQUARE	Deferred Retirement Trust		Employee 77% - District 23%	1,387.69
39078	CALPERS - UAL FOR CLASSIC MEMBERS	Annual Lump Sum Prepayment - Classic			214,383.00
39079	CALPERS - UAL FOR PEPRA MEMBERS	Annual Lump Sum Prepayment - Pepra			2,570.00
		Sub-Total for Full-Time & Trustee Payroll Taxes & Benefits			\$425,473.80
39080	ZACKARY BARRAGAN	Laboratory Tech I			1,506.09
39081	DANNY CADENA	Vector Control Tech I			1,356.57
39082	JOSE CARRERA	Laboratory Tech I			1,179.72
39083	CATHERINE CONROY	Laboratory Tech I			1,255.59
39084	ALYSIA DAVIS	Vector Control Tech I			1,600.65
39085	ARTURO GARCIA	Vector Control Tech I			1,024.40
39086	JOEY GASCA	Vector Control Tech I			1,194.21
39087	JAKE MALDONADO	Laboratory Tech I			1,505.67
39088	MARCO MARTINEZ	Vector Control Tech II			1,864.32
39089	JOSHUA MITCHELL	Laboratory Tech I			1,179.71
39090	ISAIAH ORTANEZ	Laboratory Tech I			717.76

20004	TIMOTHY PARK	Laboratory Tash I			1,179.71
39091		Laboratory Tech I			
39092	RAFAEL RAMIREZ	Vector Control Tech I			1,189.66
39093	ADRIANA RODRIGUEZ	Laboratory Tech I			520.11
39094	BRENDA RODRIGUEZ-VIVAR	Laboratory Tech II		1,679.72	
39095	LISA SALGADO	Vector Control Tech I			1,481.85
39096	ADRIAN SIFUENTES	Vector Control Tech I			1,329.92
39097	RYAN SPRATT	Vector Control Tech I			1,394.52
39098	ISAEL VENEGAS-ALAMO	Vector Control Tech I			1,193.19
39099	KORY WILSON	Vector Control Tech I			1,328.52
39100	DELTA VECTOR CONTROL DIST - EFTPS	Social Security/ Medicare/ Federal Income Tax			6,448.36
39101	DELTA VECTOR CONTROL DIST - EFTPS	State Income Tax			533.17
39102	DELTA VECTOR CONTROL DIST - EFTPS	CalPERS Retirement			1,617.27
		Sub-Total Seasonal Payroll Taxes & Benefits			34,280.69
		Total Full Time, Trustee & Seasonal Payroll Taxes & Benefits			\$459,754.49
39103	AMERICAN INC.	Quarterly Filter Change		Building Maintenance	1,195.00
39104	CENTRAL VALLEY BUSINESS FORMS	Conlin Reis, General Manager Business Cards and District Business Cards		Office Supplies	127.49
39105	COMCAST	Internet		Telephones/Cellphones	350.61
39106	EMD NETWORKING				4,744.00
		Total Care Monthly Billing - July	3916.35	Maint. Contract	
		Upgrade to Windows 11 - New Lab Laptops		Office Supplies	
		VOIP Phone System		Telephones/Cellphones	
39107	ENTERPRISE	Leased Vehicles		Capital - Vehicles	1,938.32
39108	FED EX	Prepaid Billing - Laptop and Cellphone Return		Misc. Expense	41.57
39109	FRESNO OXYGEN	Dry Ice for traps		Lab Assessment	848.48
39110	LINXUP	GPS		GPS	945.57
39111	MARIO SANCHEZ	ASE Exam Reimbursement		Continuing Education	146.00
39112	MARIO GARCHEZ	2024-2025 Corporate Membership Dues		Dues	11,500.00
39113	PACIFIC WEST CONTROLS	Monthly Service Contract		Maint. Contract	250.00
39114	SO CAL EDISON	Utilities		Utilities	6.712.48
39114	UNIFIRST	Junites		Ounties	1,385.17
39113		Uniforms	454 40	Uniforms	1,305.17
		Janitorial		Janitorial	
39116	VALLEY PACIFIC	Fuel	930.77	Fuel	4,659.82
39117	WEST COAST BIOTECH	Waste Disposal		Lab Supplies	155.00
		Total Board Order # 1			\$494,754.00
					ψτστ, 1 34.00
<b></b>					
I				1	

### 6. Staff Presentations

### 7. Manager's Report

## **DELTA MOSQUITO & VECTOR CONTROL DISTRICT**

Conlin Reis General Manager

Paul Harlien Operations Program Manager

Andrea Troupin Scientific Program Manager

Mary Ellen Gomez Administrative Assistant 1737 West Houston Avenue, Visalia, California, 93291 Phone: (559) 732-8606 Toll Free: (877) 732-8606 Fax: (559) 732-7441



Erick Arriaga Community Education & Outreach Coordinator

Rick Alvarez Vector Control Supervisor

Bryan Ruiz Supervisor Assistant

Bryan Ferguson *Foreman* 

### **REPORT OF THE MANAGER JUNE 2024**

### I. Water and Weather

The Delta Mosquito and Vector Control District (DMVCD) Weather Station reported an average high temperature of 96.9°F with an average low of 67.0°F and 0.0 inches of rainfall as of June 30, 2024. The National Oceanic and Atmospheric Administration 1991-2020 seasonal averages for high and low temperatures in June were 88.8°F and 61.5°F respectively, with average rainfall of 0.13 inches.

Water storage at Pine Flat Reservoir decreased to 885,330 acre-feet as of June 30, 2024. Pine Flat Reservoir's water inflow rate is 2,371 cubic feet per second (CFS) while its release is 6,602 CFS. The Lake Kaweah Reservoir storage decreased to 123,156 acre-feet. Lake Kaweah's water inflow rate is 2,518 CFS and its release increased to 2,549 CFS as of June 30, 2024.

### II. Narrative

Field operations received 96 mosquito service requests and 7 fish requests in June 2024. The service request inspections consisted of a Vector Control Technician inspecting the reported property for any standing water that mosquito larvae could breed in. During this time, the technician used an integrated vector management (IVM) strategy to reduce any mosquito-breeding found. In addition, technicians used this opportunity to educate residents on mosquito-breeding prevention and how to protect themselves from mosquito bites and mosquito-borne diseases.

### Operations

All the field techs are hard at work, we have moved the start time to 6:00 AM so they can get the heavy spraying done before the heat gets too hot to spray. The Urban Techs are coming in at 7:00 am to avoid the heat as much as possible and not be too early to contact residents. We are still working on swimming pools and storm drains.

The Ford Maverick has been put in service and so far, everyone likes it. Currently, we have all the fleet up and running. We do have some small repairs like AC, lights, and spray pumps. Bryan Ferguson did a drone treatment at the Ivanhoe sewer farm.

### Outreach

This month, The DMVCD took part in 6 outreach events: two Wednesday morning farmers markets in Visalia and one Saturday morning farmers market in Visalia. We attended three events of the six-event series for the Dinuba Farmers Market. New tailgate graphics have been ordered; they are expected to be completed and installed as soon as the new vehicles arrive. Staff shirts were ordered for full-time employees; they're expected to arrive within the first two weeks of July. The employee of the quarter plaque for the second quarter of 2024 has been updated and a certificate created.

The first inspections for the annual aerial pool list are complete; we had fewer issues than prior years. Any issues encountered this season will be considered for next year's annual pool inspection project.

### Laboratory

Routine mosquito trapping continued through June 2024. Traps were set throughout the District in Goshen, Traver, London, Delft Colony, Dinuba, Orosi, Seville, Kings River area, Elderwood, Woodlake, Lemon Cove, Ivanhoe, Exeter, Farmersville, and Visalia.

A total of 39,697 mosquitoes were collected across 1,674 trap nights. The District-wide average of 23.7 mosquitoes per trap night for June was 27.0% lower than the previous 5-year average of 32.5 and 65.1% lower than the 2023 average of 68.0 mosquitoes per trap night for the same time. The top 3 mosquito species caught in June were *Culex quinquefasciatus* at 67.4%, followed by *Ae. aegypti* at 23.2%, and *Cx. stigmatosoma at* 4.4%. Compared to June of 2023, the top 3 mosquito species caught were *Cx.quinquefasciatus* at 59.8%, followed by *Cx. tarsalis* at 34.0% and *Ae. aegypti* at 3.7%. Arbovirus testing for West Nile virus (WNV), St. Louis Encephalitis virus (SLEV), and Western equine encephalitis virus (WEEV) continued through June. As of June 30, 2024, DMVCD tested all 743 mosquito samples collected in June. One sample was positive for St. Louis Encephalitis virus and was collected in north Visalia. Three dead bird brain samples were collected in June and have all tested negative for WNV, SLEV, and WEEV. No kissing bugs have been reported to the district so far in June.

Technicians brought in 18 mosquito larvae samples to the laboratory for species identification in June 2024. There were 42 larval samples collected in June 2023.

There were 35 technician and homeowner mosquitofish requests in June 2024 to distribute 1,004 fish. In June 2023, there were 40 technician and homeowner mosquitofish requests to distribute 1,712 fish. Approximately 1,265 mosquitofish fry were produced by June 30, 2024.

Routine laboratory maintenance continued during June 2024. Monthly safety checks for fire extinguishers and emergency lights were performed as well as weekly safety showers and eyewash inspections.

The laboratory staff tested 59 mosquito samples from Tulare Mosquito Abatement District (MAD) in June 2024.

There were 96 service requests in June 2024.

2024	Mosquitofish	Inspection	Mosquito	Source	Other	Total
January	0	2	2	1	0	5
February	3	5	1	12	0	21
March	5	17	6	5	0	33
April	4	25	1	6	0	36
Мау	15	110	0acr	0	0	125
June	7	15	51	5	1	79
Total	34	174	61	29	1	299

### 2024 Service Request Summary

### III. Vector and Disease Surveillance

Delta MVCD Summaries:

**Humans:** No new human cases reported to DMVCD in June 2024. In 2023, there were 3 confirmed human cases of WNV and SLEV co-infection, 1 probable WNV and SLEV co-infection, 1 confirmed WNV case, and 6 probable WNV cases. There was 1 potential WNV death in 2024 but the patient contracted WNV in 2023. It is unclear whether the patient had other co-morbidities.

Birds: 3 dead birds were reported to DMVCD in June 2024 and all 3 tested negative.

**Mosquitoes**: 743 samples were collected and tested in June 2024. There is 1 SLEV positive. In 2023, 7,082 mosquito samples were tested with 473 positive WNV samples, 172 SLEV positive samples, and 47 positives for both WNV and SLEV.

### State Surveillance:

**Humans:** As of June 30, 2024, 0 human cases of WNV from 0 counties and 0 cases of SLEV from 0 counties have been reported.

Birds: As of June 30, 2024, 61 dead birds from 10 counties tested positive.

**Mosquitoes:** As of June 30, 2024, 203 mosquito samples from 14 counties have tested positive for WNV. Also, 9 samples from 3 counties have tested positive for SLEV.

### V1. Expenditures & Revenues - 2023/24

Total Budget \$5,068,115.26

EXPENDITURES – July 1, 2023, to June 30, 2024

Salaries	\$2,912,306.18
Services & Supplies	\$1,236,040.76
Tax Admin Fee	\$75,283.00
Capital	\$93,907.20
Long-Term Debt	\$117,828.06
TOTAL EXPENDITURES	\$4,435,365.20

REVENUE RECEIVED - July 1, 2023- June 30, 2024

July	\$75.00
August	\$39,702.16
September	\$31,916.62
October	\$7,591.00
November	\$523.77
December	\$2,635,329.40
January	\$285,794.26
February	\$13,558.44
March	\$55,614.48
April	\$1,509,823.04
Мау	\$458,362.10
June	<mark>Unavailable</mark>
TOTAL REVENUE TO DATE	\$5,038,290.27

### IV. Timesheet Summary

Month	Available Hours	Sick Hrs Used	Total Hrs Avail. For Work	Pct. Of Hrs Avail for Work
July	6,048	129	5,919	97.86
August	6,440	320.4	6,119.60	95.02
September	5,376	150.5	5,225.50	97.2

October	4,400	265.75	4,134.25	93.9
November	2,992	178	2,814	94.0
December	2,688	104	2,584	96.13
January	3,128	118	3,010	96.22
February	2,432	95.25	2,336.75	96.08
March	2,856	163	2,693	94.3
April	5,632	84.5	5,547.5	98.5
Мау	5,984	126	5,858	97.89
June	5,600	319.5	5,280.50	94.3

The District has a vacation policy that requires 24-hour notice to ensure the operational integrity of the workforce. Sick leave for doctor, dentist and/ or family medical necessity also requires advance notice-in so much as it is possible. Illness is unplanned and therefore unscheduled. Attendance records for the current year are shown in the table.

\*\* June expenditures and revenue are not available as of 7/5/2024

### 8. Adoption of CalPERS 457 Program



July Action Items: Background Information for the Board:

### Item 8, Adoption of CalPERS 457 Program:

### Introduction

The Delta Mosquito and Vector Control District currently offers 457 contributions under Mission Square; however, the plan has seen low utilization among employees. To enhance the retirement planning options for our staff, it is proposed that the District adopt the CaIPERS 457 program, which includes the optional loan provision and Roth component.

Key benefits of the CalPERS 457 program include:

- 1. Integration with Existing CalPERS Pension Plan: Employees can manage and view their 457 plan alongside their current CalPERS pension plan, providing a streamlined and cohesive approach to retirement planning.
- Loan Option: The program offers a loan provision that allows participants to borrow against their 457 plan balance without incurring the typical 457 withdrawal penalties, thereby providing flexibility and financial security in times of need.
- 3. Roth Component: The Roth option allows for after-tax contributions, providing employees with more flexibility in their retirement savings strategy and potential tax benefits.
- 4. Additional Option: This new program will not replace the current 457 plan under Mission Square but will provide an additional option for employees, giving them more choice in their retirement planning.
- 5. Support and Guidance: CalPERS offers advice and direction to encourage enrollment and utilization, helping employees make informed decisions about their retirement savings.

Importantly, the CaIPERS 457 program comes at no additional cost to the employer and is designed to be easily managed, ensuring a smooth transition and minimal administrative burden. This program's adoption is expected to increase participation and utilization, thereby enhancing the overall retirement readiness of our employees.

**Recommended Action:** I recommend the Board adopt the CaIPERS 457 program, which entails authorizing the General Manager and administrative staff to request addition of the 457 program to the current CaIPERS contract, approval of an enacting resolution, and authorization to sign the employer agreement.

Sincerely,

Conlin Reis, General Manager



CalPERS 457 Plan

# Your choice, simplified. Employer Welcome Kit

# Welcome to the CalPERS 457 Plan

You are introducing an exciting opportunity to your employees to invest for their future. Enclosed is all the information you need to allow your employees to participate in the Plan. If you have questions, be sure to contact your dedicated Account Manager.



### Participant Login Website

# calpers.voya.com 800-260-0659

### Participants can:

- Access information and manage their account online or through the Plan Information Line
- View, download, and print account statements
- Change investment selections among various the options available within the Plan
- Set up automatic account rebalancing to reallocate Plan balances to specific percentages
- Redistribute their account balance across multiple funds in one simple transaction
- View and update beneficiary information
- Download forms, utilize retirement planning resources, and stay up-to-date on Plan rules



### **Employer Resource Center**

# calpers-sip.com 800-696-3907

Agencies can:

- Get information and guidelines for submitting payroll information to the Plan
- · Access up-to-date information on Plan rules and enhancements
- Download participant forms and educational material for to help employees enroll and manage their account



# EMPLOYEE RETENTION IDEA #457: add this retirement plan

Like all employers, you know it can be a challenge to reward and retain the best workers. But did you know that's among the many reasons why more than 800 public agencies and schools throughout California offer the CalPERS 457 Plan?

## What's in it for **you**?

- A new reason for your agency's employees to feel good about their jobs
- No cost to your agency
- When you say "yes" to the CalPERS 457 Plan, we take care of the rest

### To help you manage your fiduciary responsibility, CalPERS:

- acts as the trustee of your employees' deferred compensation assets
- selects the investment options
- monitors contribution limits
- manages service levels with administrative service providers
- leverages existing payroll deduction feeds into myCalPERS to receive your employees' 457 Plan deferrals and loan repayments

We are a not-for-profit public entity focused on educating your employees, not selling investment products.

Funds are managed internally by our in-house investment staff and external institutional investment managers under contract to CalPERS.

We offer value and fair pricing. To keep fees as low as possible, CalPERS does not engage in "revenue sharing" or "administrative expense reimbursement." Plan administration, recordkeeping and investment fees are all included in a simple and fair fee structure.

## Find out more today

Visit calpers-sip.com

## Call 800-696-3907

Monday - Friday, 8:00 a.m. to 5:00 p.m. (PT)

## What's in it for your employees?

- The convenience of saving for retirement through payroll deduction
- A very competitively priced retirement plan, with total fund expenses of 0.25% 0.38% for the core investment options
- Access to a proprietary lineup of core institutional investments managed by CalPERS staff and professional managers under contract to CalPERS, including Target Retirement Date Funds specifically designed for public employees based on the latest research in portfolio theory and behavioral finance
- A Schwab Personal Choice Retirement Account<sup>®</sup> (PCRA) self-managed account brokerage option
- State-of-the-art services

### 24/7 account management

- Online at calpers.voya.com
- Toll free telephone at 800-260-0659
- On the go with the **Voya Retire** app

**The latest in online retirement planning,** investing and educational tools, including myOrangeMoney<sup>®</sup>, at **calpers.voya.com** 

**Dedicated Account Managers** lead group educational sessions and meet one-on-one with employees. They are easy to reach at **888-713-8244** or **calpers457.timetap.com** 

**Recordkeeping by Voya Financial**<sup>®</sup>, one of the nation's largest defined contribution plan recordkeepers, serving millions of participants

Information from registered Plan Service Representatives is for educational purposes only and is not legal, tax or investment advice. Local Plan Service Representatives are registered representatives of Voya Financial Advisors, Inc., member SIPC (VFA). Plan administration services provided by Voya Institutional Plan Services, LLC.



## Employer Adoption Checklist Coversheet

Below is a listing of documents that are required for the adoption of this Plan. When submitting the executed documents, please fill out and include this **Employer Adoption Checklist** as a coversheet to the documents listed below.

If you have any questions regarding this list or the enclosed forms, please contact us at 800-696-3907.

Agency Name	
	CalPERS ID
The following documents are included in this packet:	REQUIRED FORMS     Board Approved Resolution (with original signatures)     Employer Adoption Agreement (with original signatures)
	New Employer Information Sheet     Optional Provision
	Employer Loan Provision

#### Please mail your documentation to one of the following addresses:

Standard Delivery: CalPERS 457 Plan P.O. Box 942713 Sacramento, CA 94229-2713 **Overnight Delivery:** CalPERS 457 Plan 400 Q Street Sacramento, CA 95811

Thank you for considering the CalPERS 457 Plan as a supplemental savings option for your employees.



## Follow these simple steps to adopt the CalPERS 457 Plan today!

STEP 1:	Request your governing body to adopt the CalPERS 457 Plan using the <i>Model Resolution</i> enclosed in this kit.				
STEP 2:	Execute a copy of the enclosed <i>Employer Adoption Agreement</i>				
STEP 3:	Complete the <b>New Employer Information Sheet</b>				
STEP 4:	<ul> <li>Optional provision:</li> <li>Employer Loan Provision: If you would like to offer participants the ability to borrow from their plan accounts, then complete the Employer Loan Provision.</li> </ul>				
STEP 5:	Complete the <b>Employer Adoption Checklist</b> , encloand mail by standard or overnight mail.	ose the appropriate original fully executed documentation,			
	Standard Delivery: (Standard Mail)	Overnight Delivery:			
	CalPERS 457 Plan	CalPERS 457 Plan			
	P.O. Box 942713	400 Q Street			
	Sacramento, CA 94229-2713 Sacramento, CA 95811				

If you have any questions about adopting the CalPERS 457 Plan, contact us at 800-696-3907.

#### Upon receipt of the executed documents, the Administrator, will:

- Sign the Adoption Agreement accepting the Employer into the CalPERS 457 Plan
- Assign an *Employer Plan ID* to confirm participation in the Plan
- Send a **Welcome Kit** that includes:
  - Confirmation of participation in the Plan
  - Instructions of when to begin submitting enrollments and submission of payroll contributions file and payments through myCalPERS
  - Copy of all fully executed official documents
- An Account Manager will contact the Employer to schedule a **New Employer Orientation** and **Employee Presentations**

**NOTE:** The Employer Plan ID Number must be provided on all subsequent transaction requests submitted to the Administrator for processing.



## **Resolution Approving Adoption of CalPERS 457 Plan**

WHEREAS, \_\_\_\_\_\_(*Employer*) desires to establish a[n additional] deferred compensation plan for the benefit of its employees; and

WHEREAS, the Board of Administration ( the "Board") of the California Public Employees' Retirement System ("CalPERS") has established the CalPERS 457 Plan (the "CalPERS 457 Plan") which may be adopted by a governmental employer the employees of which are public employees; and

WHEREAS, \_\_\_\_\_(*Employer*) believes that the CalPERS 457 Plan and the investment options available thereunder will provide valuable benefits to its employees; and

WHEREAS, the Board has appointed Voya Financial<sup>®</sup> (the Plan Recordkeeper) to perform recordkeeping and administrative services under the CalPERS 457 Plan and to act as the Board's agent in all matters relating to the administration of the CalPERS 457 Plan;

NOW, THEREFORE, BE IT RESOLVED that	<i>(Employer)</i> adopts the CalPERS 457 Plan for the benefit of	
its employees and authorizes and directs the	(Title of the Authorized Member) to execute the attached	
adoption agreement on behalf of	_(Employer) and to provide CalPERS or any successor agent duly	
appointed by the Board with such information and cooperation as may be needed on an ongoing basis in the administration of		
the CalPERS 457 Plan. A copy of this resolution, the agreement, a	and any attachments thereto shall be on file in the office	
of (Office of Record).		

Passed and adopted as a resolution of the (Au	thorized Member of the Employer), at a meeting held
on (Date, if a	pplicable).

\_\_\_\_\_ (Board Chair or Authorized Member Signature)

\_\_\_\_\_ (Title of Authorized Member)

Attest \_\_\_\_\_ (Employer Seal)



### **Employer Adoption Agreement**

The employer identified below (the "Employer") adopts the CalPERS 457 Plan (the "CalPERS 457 Plan" or the "Plan") for the benefit of its employees and agrees to be bound by and subject to the terms of the Plan, as it may be amended from time to time. The Employer further agrees and represents as follows:

- 1. The Employer is a political subdivision of the State of California and is eligible to adopt the Plan.
- 2. The Employer has duly adopted a resolution (copy attached) or taken such other official action as required for its lawful adoption and implementation of the Plan and has authorized the undersigned to execute this Agreement on its behalf.
- 3. The Employer has received and has had the opportunity to review the following documents and information:
  - The Plan document;
  - A description of the optional provisions of the Plan;
  - A description of the investment options available to Plan participants and historical performance data for those investment options;
  - A complete description of fees and expenses that will or may be charged to Plan participants including, but not limited to, investment fees and administrative expenses; and
  - T he Enrollment Kit for eligible employees, which includes forms and information for employees to participate in the Plan.

#### **Contributions**

- 4. The Employer understands that its employees will have the opportunity to defer their own compensation by designating an amount or percentage to be withheld from each paycheck and contributed to the Plan on the employee's behalf.
- 5. The Employer understands that the Plan must be made available to all employees and agrees to offer all employees the opportunity to participate in the Plan.
- 6. The Employer understands that the Plan cannot be made available to individuals who are not the Employer's common law employees and agrees not to offer such individuals the opportunity to participate in the Plan.
- 7. The Employer understands that each employee's deferrals under the Plan and any other eligible deferred compensation plan maintained by the Employer are subject to certain limits imposed by the Plan and the federal tax code. The Employer agrees to limit employees' deferrals under all plans maintained by the Employer to amounts that do not exceed applicable limitations.
- 8. The Employer agrees to deduct deferral amounts from employees' salaries and wages in accordance with the employees' elections, to remit all amounts deducted to the Plan as soon as reasonably practicable after such amounts are withheld, and to accurately report the amounts remitted.
- 9. The Employer understands and agrees that all amounts deferred under the Plan shall be 100% vested and shall be deposited in the Public Employees' Deferred Compensation Fund (the "Fund"), a trust established to hold such amounts, for the exclusive benefit of participants and their beneficiaries. The Employer shall have no right to Fund assets or to sell, redeem, or otherwise liquidate Fund assets, except as provided Plan section 10.6.

#### Investments

- 10. The Employer understands and agrees that employees who defer compensation under the Plan will have the right to direct the investment of their individual Plan accounts by choosing among the investment options selected by the CalPERS Board of Administration (the "Board") and offered under the Plan. The Employer further understands and agrees that any employee who does not provide timely directions for investing his or her account will be deemed to have selected the Plan's default investment. The Plan's default investment is currently the CalPERS Target Retirement Fund designated for an employee, based on his or her expected retirement date. The Employer understands and agrees that the Board, in its sole discretion, may add, eliminate, or consolidate investment options offered under the Plan, including the Plan's default investment option.
- 11. The Employer further understands and agrees that certain fees are charged to Plan participants for investment and administration expenses, and that such fees will be offset against investment returns or deducted from participants' Plan accounts periodically.

#### **Administration**

- 12. The Employer understands and agrees that, except as specifically set forth in the Plan, the administration of the Plan and Fund is subject to the exclusive control of the Board and that the Board has the authority to retain third parties to provide investment services, record keeping, accounting, or other services for the Plan.
- 13. The Employer agrees to assist and cooperate in providing Plan information to employees and to follow administrative procedures established by the Board or its designee(s) from time to time.
- 14. The Employer has completed the attached New Employer Data Sheet, which is incorporated by reference. The Employer hereby certifies that all information provided in connection with its adoption of the Plan is true and accurate.
- 15. The Employer understands and agrees that the Board has retained the power and authority to amend the Plan from time to time, subject to limitations set forth in the California Government Code and the Plan. The Employer may not amend the Plan.
- 16. The Employer understands and agrees that its participation in the Plan may be terminated by the Employer or by the Board upon sixty (60) days advance written notice. Upon termination, all amounts held for participants will continue to be held in the Fund for the exclusive benefit of participants and their beneficiaries, except for distributions or transfers permitted under the Plan terms.

Name of Employer:		
By:	Title:	
Date:		

# Accepted by CalPERS (or an agent duly appointed by the Board) on behalf of the Board of Administration of the California Public Employees' Retirement System.

To be completed by CalPERS	Signature:	
by CalPERS	Print Name:	
	Date:	



## New Employer Information Sheet

### I. General Information

		Number of Employees
Employee Tax Identification Number	Fiscal Year	End Date / / /
Employer Address		
City / State / Zip		
Plan Administrator	Email	Telephone
Payroll Contact	Email	Telephone
	vider (new enrollments only) ve Plan Provider and convert assets — See Sectio	on III Asset Transfer Information
II. Contribution Information		
II. Contribution Information         Frequency of Payroll Deductions       Wee	skly □Bi-weekly □Semi Monthly □Monthly	
Frequency of Payroll Deductions		ria my∣CalPERS

Make check payable to CalPERS 457 Plan and include Plan ID #, and submit by standard mail or overnight.

### Please mail your documentation to one of the following addresses:

Standard Delivery: (Standard Mail)		
CalPERS 457 Plan		
P.O. Box 942713		
Sacramento, CA 94229-2713		

**Overnight Delivery:** CalPERS 457 Plan 400 Q Street Sacramento, CA 95811



## New Employer Information Sheet

### III. Asset Transfer Information

### (Complete ONLY if conducting a conversion into the CalPERS 457 Plan)

Former Plan Provider		Total Number of Employees
Address		
City / State / Zip		
Payroll Contact	Email	Telephone
IV. Signatures		
Print Name:		
Title		
Employee Signature:		
Date:		

New Employer Plan Number assigned by CalPERS: 45 \_\_\_\_\_



# Loan Feature

Planning for the unexpected can be difficult. Before you decide to tap into your CalPERS 457 Plan account, make sure you understand how a loan could impact your retirement savings.

### Who may apply for a loan?

Any participant, whose Agency has adopted the Loan Provision, may take a loan from their account with the CalPERS 457 Plan (the "Plan"). You will need to contact your Agency or the Plan Information Line **800-260-0659** to learn if your Agency has adopted the Loan Provision.

#### What are the fees associated with applying for a loan?

- The charge is \$50 per loan application paid by the participant. This fee is deducted from your Plan account.
- There is an annual maintenance fee of \$35.00, assessed on a quarterly basis as \$8.75.

The following may help you in determining if requesting a loan is really the best method for you to meet those unexpected expenses:

## Taking a loan from the Plan

### How do you apply for a loan?

As a participant in the Plan, you may apply for a loan over the Plan Information Line at **800-260-0659** or online at **calpers.voya.com.** 

### What happens if and when the loan is approved?

Once the application is submitted and approved the following will occur:

- You will receive the loan check along with a Truth-in-Lending Disclosure Statement.
- The promissory note and security agreement will be printed on the back of the loan check.
- The notice will contain the amount financed, the finance charge, the loan's annual percentage rate, the repayment procedure, the security interest and a copy for you to keep for your records.

### What are the minimum and maximum loan amounts?

The minimum loan amount is \$1,000.

The maximum loan amount is the lesser of:

- 50% of your vested account balance as of the Valuation Date of the loan or, \$50,000
  - The Valuation Date is the business day immediately preceding the date on which the loan is approved
- This amount is then reduced by your highest outstanding loan balance, if any, over the last 12 months.

### How is the maximum loan amount calculated?

The IRS Limits the amount you may take from multiple plans for loans. For the purpose of determining the maximum loan amount available to you, any loan from any other plan maintained by a participating employer will be treated as if it were a loan made from this Plan. The maximum loan amount is the lesser of: (1) 50 percent of your vested account balance as of the Valuation Date immediately preceding the date on which the loan is approved, or (2) \$50,000, less the highest outstanding loan balance over the last 12 months. By submitting this loan you are stating that you are in compliance with these regulations.

### How is the money taken out of my Plan investment offerings?

Loan disbursements will be taken pro rata across all the money sources, including any Roth balances, and investment funds in your account, excluding SMA Funds.

## Will I have to pay taxes on the loan amount as if it was a distribution from my account?

No. Amounts borrowed through the Plan are not taxable distributions and are not subject to federal withholding taxes as long as the loan is repaid in full.

#### Will I have to pay interest on my loan?

Yes, you will pay interest on your loan at a rate of the Prime Rate plus one percent, as printed in the Wall Street Journal on the last business day of the prior month. These interest payments will go back into your account; however, you will not be able to deduct this interest on your income tax return.

 The Service Members Civil Relief Act (an update to the Soldiers' and Sailors' Civil Relief Act of 1940) imposes a 6% maximum limit on the interest rate charged to military service members for loans during the duration of active military service. For more information, contact a Participant Service Representative at 800-260-0569, Monday through Friday, 6:00 a.m. to 5:00 p.m. Pacific Time, except on New York Stock Exchange holidays.

### How a loan may affect your savings

Pros	Cons
You have access to the money in your Plan account.	The amount of the interest may be less than the money you would have earned had you invested the loan balance.
You pay yourself back with interest that may be lower than a bank interest rate.	Your loan repayments are made with after-tax dollars.
No credit check.	Many participants decrease or stop contributions while paying back a loan which negatively impacts your retirement savings.

### Requesting a Loan

### There are two ways to apply for a loan:

1 Log on to the CalPERS website at **calpers.voya.com** and select your Plan.

- Under the *My Account* tab, click on *Loans*, then *Request a Loan*.
- You will be required to provide your password to request a loan.
- 2 Call the CalPERS Plan Information Line at **800-260-0659**, and speak with a Participant Service Representative. Representatives are available Monday through Friday, 6:00 a.m. to 5:00 p.m. Pacific Time, except on New York Stock Exchange holidays.

Upon approval, your loan application will be processed and a check will be mailed usually within 2 business days.

### Loan Repayments

### How long may I take to repay the loan?

The minimum loan period is 1 year and the maximum loan period is 5 years.

### How do I repay the loan?

You will repay yourself with interest through payroll deductions that are automatically deducted from your employer's payroll system.

- Payments will begin with the regular payroll beginning the second month, or as soon thereafter as is administratively practicable.
- Payroll contributions and payments, including after-tax repayments, will be submitted through the my|CalPERS system.
- Terminated participants can continue to make payments on an outstanding loan balance via reverse ACH. If a terminated participant elects the option to continue making payments, the loan will be amortized to a monthly payment. Please call the Participant Information Line to set up this option.

#### How are payments applied to my account?

Loan repayments will be reinvested according to your current investment elections.

#### May I payoff the loan in full?

Yes, you may prepay your loan in full at any time by paying the outstanding loan balance by cashier's check or certified check made payable to: CalPERS 457 Plan.

• Partial payments are not allowed.

#### What if I am called to Military Duty?

If you are called to military duty, loan repayment and the default process will be suspended. The loan repayment period is extended for the period of time you are on active military duty under the Uniformed Services Employment and Reemployment Rights Act of 1994 (USERRA).

#### What if I am on an approved unpaid leave of absence (LOA)?

Once an unpaid LOA is verified by your employer, loan payments will be suspended for up to 12 months from your leave date, but not to exceed your originally scheduled loan payoff date. When you return to work, your loan will be re-amortized to a new loan repayment amount that includes your missed payments. Please call the Participant Information Line to notify Voya of your leave and your return to work.

## Under what circumstances may my loan be immediately due and payable?

 Upon death, insufficient paycheck funds, retirement or termination of employment, your outstanding loan balance will be immediately due and payable. Terminated participants also have the option of continuing loan payments as an amortized monthly amount instead.

### Loan Restrictions

### How many loans may be taken out?

- For all CalPERS 457 Plans, the maximum number of loans permitted at one time is one.
- No new loan may be taken until the outstanding loan is paid off.
- Loan refinancing is not allowed.
- Loans can not be taken on the self-managed Personal Choice Retirement Account's investments.

## If I have a loan outstanding may I transfer monies between participating employers' plans?

- If you have an outstanding loan with the CaIPERS 457 Plan and request a transfer of your account to another CaIPERS 457 Plan maintained by a participating employer, CaIPERS will process the transfer of your unencumbered account balance less the outstanding loan balance payable to the CaIPERS 457 Plan.
  - Further transfer requests will not be processed until your loan with the CalPERS 457 Plan has been repaid in full.
- If you have an outstanding loan from any other plan provider maintained by a CalPERS 457 participating employer, CalPERS will accept a transfer of your unencumbered account balance from the other plan, less the amount of the outstanding loan balance payable to the other plan.

## May I use something besides my Plan account as collateral for a loan?

No. A loan may only be secured by an interest in your vested account pre-tax balance and Roth balance, if available.

### Loan Default

#### What happens if I fail to make a scheduled payment?

CalPERS will treat a loan in default if any scheduled repayment remains unpaid after the expiration of the maximum grace period — the last day of the calendar quarter following the calendar quarter in which the required repayment was due, or if there is outstanding principal existing on the loan after the last scheduled repayment date.

#### What happens if I enter into default on my loan?

If your loan is defaulted, your vested account balance will be offset by the outstanding loan balance to the extent that a distribution from your account is permissible under the Plan.

• The distribution made for the loan repayment will be reported as earned income and a 1099R will be issued.

# Employer Loan Provision Form

#### I. General Information

Agency/Employer Plan Number: **45** \_\_\_\_\_ \_\_\_\_

#### Employer Name

Contact Name

Email Address

#### II. Adoption of CalPERS 457 Loan Program

The Employer referenced above adopted the CalPERS 457 Plan (Plan) effective \_\_\_\_\_\_, and is a sponsor of the Plan. The individual signing below hereby certifies that he or she is duly authorized to execute this form on behalf of the Employer and that all necessary action has been taken by the Employer to authorize and approve this action. The Employer hereby elects to offer the CalPERS 457 Loan Program Option to its employees. This election may be revoked at any time; employees with outstanding loans at the time of revocation will be permitted to pay off the loans, but will not be permitted to take out new loans.

The loan provisions are set forth in Section 8 of the CalPERS 457 Deferred Compensation Plan Document. The Employer acknowledges that it has received a copy of the current Plan document and has had the opportunity to review the terms for the Loan Program. Here is a summary of the program:

- The minimum loan amount is \$1,000; the maximum loan amount is the lesser of 50% of a participant's account balance or \$50,000 (reduced by the highest balance of any loan outstanding in the last twelve months). The maximum limit is applied taking into account all loans outstanding from any retirement plan sponsored by the Employer.
- Loans must use level amortization, require payments at least quarterly, and may not have a term longer than five years.
- Interest on a loan is charged at a rate equal to prime plus 1% and the loan must be secured by the participant's account balance.
- Loan payments must be made by payroll deduction and are made on an "after-tax" basis.
- A participant may pay off the balance of an outstanding loan, in full, with no prepayment penalty. Partial payments are not accepted.
- Weekly feedback reports are provided via the Sponsor Website. These feedback reports provide information regarding any new loans issued or loans paid during the prior week. In addition, an automated email is sent to the Employer's designated contact advising when new feedback reports become available. It is the Employer's responsibility to ensure that contact information for the Employer's designated contact is current and correct. Additionally, it is the Employer's responsibility to monitor weekly feedback reports.

Telephone #

 If the Employer has multiple plan providers which permit participant loans, it is the Employer's responsibility to monitor that a participant has not exceeded the maximum loan limit as described by the IRS. (\$50,000). Please contact CaIPERS immediately if you become aware of an issue. Participants submitting online loan requests are notified of IRS loan limits at the time the request is made, however, the Employer also agrees to monitor aggregate balances of participants with loans from multiple providers.

#### III. Signature

I hereby acknowledge and agree to the terms of this form.

	/	/	
Date			

Printed Name

By

Title

Accepted by CalPERS (or an agent duly appointed by the Board) on behalf of the Board of Administration of the California Public Employees' Retirement System.

/ / Effective Date Title



SKU #	<b>#LPF-081</b>	1 - 3031	084.G.S



# **Roth Contribution Option Overview**

#### What is the Roth Contribution Option?

The CalPERS 457 Plan offers a voluntary Roth after-tax contribution option to give you more flexibility when saving for retirement. If your employer can process your Roth contribution from their payroll to the CalPERS 457 Plan, you can elect to make Roth contributions of money from your paycheck that has already been taxed. This helps you build a nest egg of tax-free income in retirement.

#### How do I make Roth contributions to the CalPERS 457 Plan?

- If you are already contributing the CalPERS 457 Plan, follow these steps:
- **1.** Confirm with your employer that they can process your Roth contribution from their payroll to the CalPERS 457 Plan.
- 2. Visit calpers457.com to download a a Participant Change Authorization Form.
- **3.** Complete the form, choosing to add or replace pre-tax savings with the Roth contribution source in the **Change Contribution Amount** section.
- **4.** Sign the **Participant Change Authorization Form** and get your employer's signature on page 2 before submitting for processing.
- **5.** Roth after-tax contributions will begin to be deducted from your net pay within 2-3 pay periods.

#### If you are not currently participating in the CalPERS 457 Plan, follow these steps:

- **1.** Confirm with your employer that they participate in the CalPERS 457 Plan and can process your Roth contributions from their payroll.
- 2. Visit calpers457.com to download and complete the Participant Enrollment Kit.
- **3.** On the *Employee New Enrollment Form*, choose the Roth contribution source in the Employee Contribution Election section.
- **4.** Sign the *Employee New Enrollment Form* and get your employer's signature on page 2 before submitting all paperwork for processing.
- **5.** Roth after-tax contributions will begin to be deducted from your net pay within 2-3 pay periods.

#### Key Benefits with CalPERS

- Invest Roth contributions using automatic payroll deductions.
- Roth contributions can be invested in the same menu of CalPERS Target Date Funds and Core Funds.
- Roth contributions to the CaIPERS 457 Plan allow you to contribute more than into an IRA you set up on your own.

#### Managing Your Account

You can access all of your CalPERS 457 Plan account information in one place.

- Register or log into your account at calpers.voya.com.
- Call **800-260-2659** and press 2 for a Participant Service Representative weekdays between 6:00 a.m. - 5:00 p.m. PT.
- Download the Voya Retire mobile app.
   Search keyword: Voya Retire at your favorite app store.

#### Schedule an Appointment

Dedicated Account Managers are available to review and discuss your retirement saving strategy. Go to <u>calpers457.timetap.com</u> or call **888-713-8244** to schedule an appointment at a date and time that is most convenient for you.

Information from registered Plan Service Representatives is for educational purposes only and is not legal, tax or investment advice. Local Plan Service Representatives are registered representatives of Voya Financial Advisors, Inc., member SIPC (VFA). Plan administration services provided by Voya Institutional Plan Services, LLC.



# Here's how the Roth contribution option works

#### My income is too high to contribute to a Roth IRA. Can I make designated Roth contributions to the CalPERS 457 Plan?

Yes, you can contribute to the designated Roth account in your CalPERS Plan even if your income is too high to be able to contribute to a Roth IRA.

## Can I contribute to both the pre-tax and Roth accounts in the CalPERS 457 Plan?

Yes. You can contribute to both accounts in the same year in any proportion you choose. The combined amount of all contributions must not exceed the annual deferral limit. Visit <u>voya.com/irslimits</u> for information on the annual deferral and catch-up contribution limits for 457 Plans.

#### Can I change or stop making voluntary Roth contributions at any time?

Yes. Please complete a *Participant Change Authorization Form* to stop or change your Roth contributions.

#### I've made Roth contributions and have changed my mind. Can I change them to pre-tax contributions instead?

No. Once you have designated contributions as Roth, you cannot later change them to pre-tax.

# Are my Roth and pre-tax contribution sources kept separate?

Yes. Designated Roth contributions are kept completely separate from your pre-tax contributions within the CalPERS 457 Plan. Your employer has established a separate account for each participant to make designated Roth contributions.

## Can I convert the pre-tax contributions I've made and their earnings to the Roth plan?

Yes. You can convert part or all of your vested plan balance, including earnings, to the designated Roth account. This is called an "in-plan Roth conversion." You can convert part or all of your pre-tax contributions, pre-tax rollover contributions from qualified former employer plans, and their earnings. Generally, the amount you convert becomes part of your gross income for the tax year in which you convert it.

Please consider all of your options and consult with your tax advisor before making any decisions to request an in-plan Roth conversion. For more information, you can also refer to the *In-Plan Roth Conversions – Frequently Asked Questions* document that accompanies the conversion request form.

You are strongly urged to consult with an accountant and/or tax advisor before making your final decision to contribute to the CalPERS Roth 457 Plan. Once Roth contributions are made, they are irrevocable and cannot be reversed. While CalPERS and Voya Financial® representatives are able to explain the options to you, they cannot tell you whether Roth contributions are appropriate for you. CalPERS and Voya<sup>®</sup> will not provide tax or legal advice. Neither this document, Employee New Enrollment Form nor the Participant Change Authorization Form represents tax or legal advice.

**CalPERS 457 Deferred Compensation Plan Document** 

# **TABLE OF CONTENTS**

Article I	- Definitions	
1.1	Account	.5
1.2	Adoption Agreement	.5
1.3	Beneficiary	.5
1.4	Board	.5
1.5	Code	.5
1.6	Deferral	.6
1.7	Deferral Agreement	.6
1.8	Eligible Deferred Compensation Plan	.6
1.9	Employee	
1.10	Employer	
1.11	Fund	
1.12	Government Code	.6
1.13	Includible Compensation	
1.14	Investment Option	
1.15	Normal Retirement Age	
1.16	Participant	
1.17	CalPERS	
1.18	Plan	
1.19	Rehired Employee	
1.20	Required Beginning Date	
1.20	Roth Elective Deferrals	
1.22	Self-Managed Account	
1.22	Severance from Employment	
1.24	Transfer	
1.25	Trustee	
1.26	Unforeseeable Emergency	
	Year	
	I - Employer Participation	
2.1	Initial Participation	
2.2	Terms of Participation	
2.2	Duration of Employer Participation.	
-	II - Employee Participation	
3.1	Eligibility	
3.2	Initial Enrollment	
3.2	Effect of Deferral Agreement	
3.4	Amendment of Deferral Agreement	
3.5	Investment Options	
3.6	Beneficiary Designation	
3.0 3.7	Additional Deferrals	
3.7	Roth Elective Deferrals	
	V - Deferral Limitations	
4.1	General Limitation.	
4.1	Age 50 Catch-up Deferral Contributions	
4.2	Special Section 457 Catch-up Limitation	
4.3 4.4	Special Rules	
4.4 4.5	Deferrals Prior To or After Severance from Employment, Including Sick, Vacation, and	<u>۱</u>
4.3		
	- 2 -	

4.6       Correction of Excess Deferrals.       14         4.7       Protection of Persons Who Serve in a Uniformed Service       14         Article V - Participant Accounts.       14         5.1       Crediting of Accounts       14         5.2       Account Balances       15         Article VI - Distributions       15         6.1       Commencement of Distributions upon Severance from Employment       15         6.2       Optional Distribution Forms       15         6.3       Deferred Commencement Date       16         6.4       In-Service Distributions From Rollover Account       16         6.4       In-Service Distributions From Rollover Account       16         6.5       Cash-outs and Combined Payments       16         6.6       Unforesceable Emergency Distributions       16         6.7       Death Benefits       16         6.8       Uniformed Service Member Distributions       18         6.10       Distribution Date       18         6.11       Transfers and In-Plan Conversions       20         7.1       Acceptance of Transfers       20         7.2       Eligible Rollover Contributions to the Plan       21         7.3       Transfers to other Plans       21
Article V - Participant Accounts.       14         5.1       Crediting of Accounts       14         5.2       Account Balances       15         Article VI - Distributions       15         6.1       Commencement of Distributions upon Severance from Employment       15         6.2       Optional Distribution Forms       15         6.3       Deferred Commencement Date       16         6.4       In-Service Distributions From Rollover Account       16         6.4       Unforesceable Emergency Distributions       16         6.5       Cash-outs and Combined Payments       16         6.6       Unforesceable Emergency Distributions       16         6.7       Death Benefits       16         6.8       Uniformed Service Member Distributions       18         6.9       Latest Distribution Date       18         6.10       Distributions/Rollovers from Fund       19         Article VII – Transfers and In-Plan Conversions.       20       20         7.1       Acceptance of Transfers.       20         7.2       Eligible Rollover Contributions to the Plan       21         7.3       Transfers to other Plans       21         7.4       Transfer Conditions       22      <
5.1       Crediting of Accounts       14         5.2       Account Balances       15         Article VI - Distributions       15         6.1       Commencement of Distributions upon Severance from Employment       15         6.2       Optional Distribution Forms       15         6.3       Deferred Commencement Date       16         6.4       In-Service Distributions From Rollover Account       16         6.5       Cash-outs and Combined Payments       16         6.6       Unforesceable Emergency Distributions       16         6.7       Death Benefits       16         6.8       Uniformed Service Member Distributions       18         6.10       Distributions/Rollovers from Fund       19         Article VII – Transfers and In-Plan Conversions       20       20         7.1       Acceptance of Transfers       20         7.2       Eligible Rollover Contributions to the Plan       21         7.3       Transfers to other Plans       21         7.4       Transfers       22         7.5       Permissive Service Credit Transfers       22         7.6       In-Plan Roth Conversions       22         8.1       Loans       23         8.2
5.2       Account Balances       15         Article VI - Distributions       15         6.1       Commencement of Distributions upon Severance from Employment       15         6.2       Optional Distribution Forms       15         6.3       Deferred Commencement Date       16         6.4       In-Service Distributions From Rollover Account       16         6.5       Cash-outs and Combined Payments       16         6.6       Unforeseeable Emergency Distributions       16         6.7       Death Benefits       16         6.8       Uniformed Service Member Distributions       16         6.9       Latest Distribution Date       18         6.10       Distributions/Rollovers from Fund       19         Article VII – Transfers and In-Plan Conversions       20       20         7.1       Acceptance of Transfers       20         7.2       Eligible Rollover Contributions to the Plan       21         7.3       Transfers to other Plans       21         7.4       Transfers       22         7.5       Permissive Service Credit Transfers       22         7.6       In-Plan Roth Conversions       22         8.1       Loans       23         8.2
Article VI - Distributions       15         6.1       Commencement of Distributions upon Severance from Employment       15         6.2       Optional Distribution Forms       15         6.3       Deferred Commencement Date       16         6.4       In-Service Distributions From Rollover Account       16         6.5       Cash-outs and Combined Payments       16         6.6       Unforesceable Emergency Distributions       16         6.7       Death Benefits       16         6.8       Uniformed Service Member Distributions       18         6.9       Latest Distribution Date       18         6.10       Distributions/Rollovers from Fund       19         Article VII – Transfers and In-Plan Conversions       20       20         7.1       Acceptance of Transfers       22         7.2       Eligible Rollover Contributions to the Plan       21         7.3       Transfers to other Plans       21         7.4       Transfers Service Credit Transfers       22         7.5       Permissive Service Credit Transfers       22         7.6       In-Plan Roth Conversions       23         8.1       Loans       23         8.2       Maximum Loan Amount       23
6.1       Commencement of Distributions upon Severance from Employment       15         6.2       Optional Distribution Forms       15         6.3       Deferred Commencement Date       16         6.4       In-Service Distributions From Rollover Account       16         6.4       In-Service Distributions From Rollover Account       16         6.5       Cash-outs and Combined Payments       16         6.6       Unforeseeable Emergency Distributions       16         6.7       Death Benefits       16         6.8       Uniformed Service Member Distributions       18         6.9       Latest Distribution Date       18         6.10       Distributions/Rollovers from Fund       19         Article VII – Transfers and In-Plan Conversions       20         7.1       Acceptance of Transfers       20         7.2       Eligible Rollover Contributions to the Plan       21         7.3       Transfers to other Plans       21         7.4       Transfer Conditions       22         7.5       Permissive Service Credit Transfers       22         7.6       In-Plan Roth Conversions       22         8.1       Loans       23         8.3       Terms of Loan       23
6.2       Optional Distribution Forms       15         6.3       Deferred Commencement Date       16         6.4       In-Service Distributions From Rollover Account       16         6.4       In-Service Distributions From Rollover Account       16         6.5       Cash-outs and Combined Payments       16         6.6       Unforeseeable Emergency Distributions       16         6.7       Death Benefits       16         6.8       Uniformed Service Member Distributions       18         6.9       Latest Distribution Date       18         6.10       Distributions/Rollovers from Fund       19         Article VII – Transfers and In-Plan Conversions       20         7.1       Acceptance of Transfers       20         7.2       Eligible Rollover Contributions to the Plan       21         7.3       Transfers to other Plans       21         7.4       Transfer Conditions       22         7.5       Permissive Service Credit Transfers       22         7.6       In-Plan Roth Conversions       22         8.1       Loans       23         8.2       Maximum Loan Amount       23         8.3       Terms of Loan       23         8.4
6.2       Optional Distribution Forms       15         6.3       Deferred Commencement Date       16         6.4       In-Service Distributions From Rollover Account       16         6.4       In-Service Distributions From Rollover Account       16         6.5       Cash-outs and Combined Payments       16         6.6       Unforeseeable Emergency Distributions       16         6.7       Death Benefits       16         6.8       Uniformed Service Member Distributions       18         6.9       Latest Distribution Date       18         6.10       Distributions/Rollovers from Fund       19         Article VII – Transfers and In-Plan Conversions       20         7.1       Acceptance of Transfers       20         7.2       Eligible Rollover Contributions to the Plan       21         7.3       Transfers to other Plans       21         7.4       Transfer Conditions       22         7.5       Permissive Service Credit Transfers       22         7.6       In-Plan Roth Conversions       22         8.1       Loans       23         8.2       Maximum Loan Amount       23         8.3       Terms of Loan       23         8.4
6.3       Deferred Commencement Date       16         6.4       In-Service Distributions From Rollover Account.       16         6.5       Cash-outs and Combined Payments       16         6.6       Unforesceable Emergency Distributions       16         6.7       Death Benefits       16         6.8       Uniformed Service Member Distributions       16         6.9       Latest Distribution Date       18         6.10       Distributions/Rollovers from Fund       19         Article VII – Transfers and In-Plan Conversions       20         7.1       Acceptance of Transfers       20         7.2       Eligible Rollover Contributions to the Plan       21         7.3       Transfers to other Plans       21         7.4       Transfer Conditions       22         7.5       Permissive Service Credit Transfers       22         7.6       In-Plan Roth Conversions       22         8.1       Loans       23         8.2       Maximum Loan Amount       23         8.3       Terms of Loan       23         8.4       Security for Loan; Default.       23         8.5       Repayment       24         Article IX - Participant Rights       25
6.5Cash-outs and Combined Payments166.6Unforesceable Emergency Distributions166.7Death Benefits166.8Uniformed Service Member Distributions186.9Latest Distribution Date186.10Distributions/Rollovers from Fund19Article VII – Transfers and In-Plan Conversions207.1Acceptance of Transfers207.2Eligible Rollover Contributions to the Plan217.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
6.6Unforeseeable Emergency Distributions166.7Death Benefits166.8Uniformed Service Member Distributions186.9Latest Distribution Date186.10Distributions/Rollovers from Fund19Article VII – Transfers and In-Plan Conversions207.1Acceptance of Transfers.207.2Eligible Rollover Contributions to the Plan217.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers.227.6In-Plan Roth Conversions22Article VIII – Loans238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default.238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
6.6Unforeseeable Emergency Distributions166.7Death Benefits166.8Uniformed Service Member Distributions186.9Latest Distribution Date186.10Distributions/Rollovers from Fund19Article VII – Transfers and In-Plan Conversions207.1Acceptance of Transfers207.2Eligible Rollover Contributions to the Plan217.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions227.6In-Plan Roth Conversions238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.3Nonassignability25
6.7Death Benefits166.8Uniformed Service Member Distributions186.9Latest Distribution Date186.10Distributions/Rollovers from Fund19Article VII – Transfers and In-Plan Conversions207.1Acceptance of Transfers207.2Eligible Rollover Contributions to the Plan217.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions227.6In-Plan Roth Conversions238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
6.8Uniformed Service Member Distributions186.9Latest Distribution Date186.10Distributions/Rollovers from Fund19Article VII – Transfers and In-Plan Conversions207.1Acceptance of Transfers207.2Eligible Rollover Contributions to the Plan217.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions227.6In-Plan Roth Conversions238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
6.9Latest Distribution Date186.10Distributions/Rollovers from Fund19Article VII – Transfers and In-Plan Conversions207.1Acceptance of Transfers207.2Eligible Rollover Contributions to the Plan217.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions22Article VIII – Loans238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
6.10Distributions/Rollovers from Fund19Article VII – Transfers and In-Plan Conversions207.1Acceptance of Transfers207.2Eligible Rollover Contributions to the Plan217.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions22Article VIII – Loans238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
Article VII – Transfers and In-Plan Conversions.207.1Acceptance of Transfers.207.2Eligible Rollover Contributions to the Plan217.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions22Article VIII – Loans238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
7.1Acceptance of Transfers.207.2Eligible Rollover Contributions to the Plan217.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions22Article VIII – Loans238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
7.2Eligible Rollover Contributions to the Plan217.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions22Article VIII – Loans238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
7.3Transfers to other Plans217.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions22Article VIII – Loans238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
7.4Transfer Conditions227.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions22Article VIII – Loans238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loar; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
7.5Permissive Service Credit Transfers227.6In-Plan Roth Conversions22Article VIII – Loans238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
7.6In-Plan Roth Conversions22Article VIII – Loans238.1Loans238.2Maximum Loan Amount238.3Terms of Loan238.4Security for Loan; Default238.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
Article VIII – Loans.238.1Loans.238.2Maximum Loan Amount.238.3Terms of Loan .238.4Security for Loan; Default.238.5Repayment .24Article IX - Participant Rights.259.1Participants' Interest in the Fund .259.2Benefits Based on Account Balances .259.3Nonassignability.25
8.1Loans
8.2Maximum Loan Amount.238.3Terms of Loan238.4Security for Loan; Default.238.5Repayment24Article IX - Participant Rights.259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability.25
8.3Terms of Loan238.4Security for Loan; Default.238.5Repayment24Article IX - Participant Rights.259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
8.4Security for Loan; Default.238.5Repayment
8.5Repayment24Article IX - Participant Rights259.1Participants' Interest in the Fund259.2Benefits Based on Account Balances259.3Nonassignability25
Article IX - Participant Rights
9.1Participants' Interest in the Fund
<ul> <li>9.2 Benefits Based on Account Balances</li></ul>
9.3 Nonassignability
$Z_{\tau}$ = ransies under Domestie Relations Orders
9.5 Release from Liability to Participant
9.6 Participation in Legal Proceedings
Article X - Terms of Fund Investments
10.1 Use of Fund
10.2 Administration of Fund
10.3 Investment Options
10.4 Fund Investments
10.5 Valuation and Accounting
6
•
10.6 Redemption Restrictions
10.6 Redemption Restrictions27Article XI - Administration of Plan28

11.4	Plan Expenses	29
11.5	Communications from Participants	29
11.6	Communications to Employers	29
11.7	Communications to Participants	29
11.8	Time Periods	
Article	XII - General Provisions	
12.1	Amendment	
12.2	Effect on Employment	
12.3	Binding Contract	
12.4	Supplementary Information and Procedures	
12.5	Incompetence of Payee	
12.6	Inability to Locate Beneficiary	31
12.7	Applicable Law	

The purpose of the California Public Employees' Deferred Compensation Plan (the "Plan") is to provide deferred compensation for California public employees that elect to participate in the Plan. This Plan is established pursuant to Sections 21670 through 21685 of the Government Code of the State of California and is intended to constitute an "eligible deferred compensation plan" within the meaning of Section 457 of the Federal Internal Revenue Code. Except as otherwise provided herein, this amendment and restatement of the Plan is effective January 1, 2020.

#### **Article I - Definitions**

The following terms when used herein shall have the following meaning:

**1.1** Account: The bookkeeping account maintained with respect to each Participant which reflects the value of the deferred compensation credited to the Participant, including the Participant's Deferrals, the earnings or loss of the Fund (net of Fund expenses) allocable to the Participant, any Transfers for the Participant's benefit, and any distributions made to the Participant or the Participant's Beneficiary.

If a Participant has more than one Beneficiary at the time of the Participant's death, then a separate Account Balance shall be maintained for each Beneficiary. The Account Balance includes any account established under Section 7.2 for rollover contributions and plan-to-plan transfers made for a Participant, the account established for a Beneficiary after a Participant's death, and any account or accounts established for an alternate payee (as defined in Section 414(p)(8) of the Code).

**1.2** Adoption Agreement: The agreement under which an Employer becomes a participating Employer under this Plan.

**1.3 Applicable Required Minimum Distribution (RMD) Age:** The Applicable RMD Age is (a) age 70-1/2 if the individual attains age 70-1/2 prior to January 1, 2020; (b) age 72 if the individual attains age 70-1/2 after December 31, 2019 and age 72 before January 1, 2023; (c) age 73 if the individual attains age 72 after December 31, 2022 and age 73 before January 1, 2033; or (d) age 75 if the individual attains age 74 after December 31, 2032. Notwithstanding the foregoing, the Applicable RMD Age will be the age prescribed by Section 401(a)(9) of the Code, including any regulations thereunder.

**1.4 Beneficiary:** The person or persons designated by the Participant to receive distributions from the Participant's Account after the Participant's death. A designated person may include, but is not limited to, one or more of the following: an individual, trust, corporation or firm, or the estate of the Participant. A designation shall be made on a Board-approved beneficiary designation form.

**1.5 Board:** The Board of Administration of CalPERS.

**1.6** Code: The Federal Internal Revenue Code of 1986, as amended from time to time.

**1.7 Deferral:** An amount credited to a Participant's Account by reason of the Participant's agreement to contribute a portion of his or her salary or wages to the Plan.

**1.8 Deferral Agreement:** The agreement between an Employer and an Employee, including any amendments thereto, which specifies the amount of Deferrals to be made by the Employee. Each Deferral Agreement or amendment thereto shall be made or confirmed in writing under procedures established by the Board.

**1.9 Eligible Deferred Compensation Plan:** An eligible governmental plan as defined in Section 1.457-2(f) of the Income Tax Regulations.

**1.10** Employee: Any individual who is a common law employee of an Employer and is a member of CalPERS or for whom the Board is otherwise authorized to administer this Plan under the Government Code. An Employee does not include an independent contractor.

**1.11 Employer:** Any political subdivision of the State of California, or any agency or instrumentality of the State of California or political subdivision of the State of California for which the Board is authorized to administer this Plan under the Government Code and that has become a participating employer under this Plan pursuant to Article 2. Where required by the context, references to the Employer shall mean the current or former Employer of the Employee or Participant.

**1.12** Fund: The Public Employees' Deferred Compensation Fund that has been established as part of the Plan pursuant to Section 21676 of the Government Code. For purposes of Section 9.1 of this Plan, the Fund shall not include the asset management and services account maintained pursuant to Section 21678 of the Government Code.

**1.13** Government Code: Those statutes of the State of California that have been codified as the Government Code.

**1.14** Includible Compensation: A Participant's compensation, as defined in Section 415(c)(3) of the Code, for services performed for the Employer. Includible Compensation shall be determined without regard to any community property laws.

**1.15 Investment Option:** One of the available alternatives for crediting investment earnings to a Participant's Account, which shall be based upon the performance of one or a combination of the investment portfolios maintained under the Fund.

**1.16** Normal Retirement Age: The age used to determine the three-year period in which a Participant may utilize the catch-up limitation under Section 4.3. A Participant may designate as his or her Normal Retirement Age an age that is not earlier than age 62 and that is not later than age 70-1/2. Notwithstanding the preceding sentence, a Participant may designate as his or her Normal Retirement Age an age that is earlier than age 62 if the Participant can demonstrate that such age is not earlier than the age the Participant will attain in the earliest Year in which the

Participant will be eligible to retire without actuarial or similar reduction under CalPERS or another retirement system.

Once a Participant has utilized the catch-up limitation under Section 4.3 or under a comparable provision of another Eligible Deferred Compensation Plan that Participant's Normal Retirement Age may not thereafter be changed. An Employer sponsoring more than one Eligible Deferred Compensation Plan may not permit a Participant to have more than one Normal Retirement Age under the Eligible Deferred Compensation Plans it sponsors.

**1.17 Participant:** Any Employee or former Employee for whom a Deferral has been credited under the Plan and for whom an Account is maintained.

1.18 CalPERS: The California Public Employees' Retirement System.

**1.19 Plan:** The California Public Employees' Deferred Compensation Plan established pursuant to Sections 21670 through 21685 of the Government Code, the terms of which are set forth in this Plan document. To the extent required under Section 457 of the Code, each Employer's participation in this Plan shall be treated as a separate plan, and each Employer's separate plan shall be deemed to include any other Eligible Deferred Compensation Plan maintained by that Employer.

**1.20** Rehired Employee: An Employee who has had a Severance from Employment with an Employer, and thereafter becomes an Employee of that Employer. An Employee shall be treated as a Rehired Employee on the date he or she becomes reemployed with the Employer.

**1.21** Required Beginning Date: April 1st of the Year following the Year of a Participant's attainment of the Applicable RMD Age or Severance from Employment, whichever is later.

**1.22 Roth Elective Deferrals:** A Deferral that is: (a) Designated irrevocably by the Participant at the time of the Deferral Agreement as a Roth Elective Deferral that is being made in lieu of all or a portion of the pre-tax elective Deferrals the Participant is otherwise eligible to make under the Plan; and (b) Treated by the Employer as includible in the Participant's income at the time the Participant would have received that amount in cash if the Participant had not made a cash or deferred election.

**1.23** Self-Managed Account: A brokerage account established by a Participant through which such Participant makes self-directed investments with respect to amounts reflected in the Participant's account.

**1.24** Severance from Employment: The date that the Employee dies, retires, or otherwise has a severance from employment with the Employer, as determined by the Employer (and taking into account guidance issued under the Code). A Rehired Employee shall no longer be treated as having a Severance from Employment upon the date of his or her rehire.

**1.25** Transfer: An amount credited to a Participant's Account by reason of a transfer from another Eligible Deferred Compensation Plan.

#### **1.26 Trustee:** The Board of Administration of CalPERS.

1.27 **Unforeseeable Emergency:** A severe financial hardship of the Participant resulting from: an illness or accident of the Participant, the Participant's spouse, the Participant's domestic partner pursuant to a state domestic relations law who is a designated primary Beneficiary or the Participant's dependent (as defined in Section 152(a) of the Code, without regard to Section 152(b)(1), (b)(2), and (d)(1)(B)); loss of the Participant's property due to casualty (including the need to rebuild a home following damage to a home not otherwise covered by homeowner's insurance, e.g., as a result of a natural disaster); the need to pay for the funeral expenses of the Participant's spouse or dependent (as defined in Section 152(a) of the Code, without regard to Section 152(b)(1), (b)(2), and (d)(1)(B); or other similar extraordinary and unforeseeable circumstances arising as a result of events beyond the control of the Participant. For example, the imminent foreclosure of or eviction from the Participant's primary residence may constitute an Unforeseeable Emergency. In addition, the need to pay for medical expenses, including nonrefundable deductibles, as well as for the cost of prescription drug medication, may constitute an Unforeseeable Emergency. A need to send a child to college or to purchase a new home shall not constitute an Unforeseeable Emergency.

#### **1.28** Year: A calendar year.

## **Article II - Employer Participation**

**2.1 Initial Participation:** This Plan shall be available to the Employees of an Employer only if the Employer has executed an Adoption Agreement and provided the Board with such information as the Board deems necessary to administer the Plan on behalf of the Employer. CalPERS, in its sole discretion, will decide whether an Employer may participate in this Plan.

**2.2** Terms of Participation: By executing an Adoption Agreement, an Employer agrees to adhere to all terms and conditions of the Plan, as amended from time to time, to invest all Deferrals and Transfers in the Fund, and to follow all administrative procedures established by the Board. Except as otherwise provided herein, the terms of this Plan shall apply on a uniform basis to all Employers participating hereunder. The Board may, in its sole discretion, upon sixty (60) days advance written notice terminate an Adoption Agreement with an Employer for participation of its Employees in the Plan.

**2.3 Duration of Employer Participation:** In the event that an Employer withdraws from participation in the Plan, all amounts credited to the Accounts of the Employer's participating Employees will continue to be held in the Fund and will be distributed in accordance with the terms of the Plan, except to the extent of any transfers from the Plan pursuant to Section 7.3.

## Article III - Employee Participation

**3.1** Eligibility: Except as provided in Section 21670 of the Government Code, all Employees of an Employer shall be eligible to participate in the Plan. Notwithstanding the above, an Employee

who is a minor may only participate in this Plan to the extent he or she may enter into an enforceable contract that cannot be disaffirmed by the minor.

**3.2 Initial Enrollment:** In order to become a Participant, an Employee must enter into a Deferral Agreement pursuant to any procedures established by the Board; provided, however, that prior to January 1, 2023, such Deferral Agreement shall become effective no earlier than the calendar month following the month in which the agreement is made. A Deferral Agreement will be given effect only if the Deferral amount elected therein satisfies whatever minimum the Board may establish, and the Employee provides all information called for on the agreement form.

**3.3** Effect of Deferral Agreement: Commencing with the effective date of an Employee's Deferral Agreement, his or her gross salary or wages shall be reduced by the Deferral amount specified in the Deferral Agreement. Deferrals shall continue to be made in such amount unless and until the Deferral Agreement is amended or the Employee has a Severance from Employment with the Employer. Subject to the limitations of Article 4, Deferrals shall not be subject to Federal or California income tax withholding and shall not be reported as gross income on the Employee's annual wage statement (Form W-2). However, Deferrals shall be subject to FICA taxation when earned.

**3.4 Amendment of Deferral Agreement:** A Participant may amend or revoke his or her Deferral Agreement at any time pursuant to any procedures established by the Board; provided, however, that prior to January 1, 2023, no change in the amount of a Participant's Deferrals will become effective until the calendar month following the month in which the Deferral Agreement is amended.

3.5 Investment Options: Upon initial enrollment in the Plan, the Participant's future Deferral will be invested in the Plan's designated "default" option for investment of contributions unless the Participant makes an affirmative investment election to the contrary prior to the date his or her Deferral is first invested. To the extent the Participant does not wish to have his or her contributions invested in the designated "default" Investment Option, the Participant may designate another Investment Option or Investment Options to which his or her Deferrals are to be allocated or elect to make investments through a Self-Managed Account. A Participant may thereafter re-allocate his or her Account balance among the available Investment Options or through a Self-Managed Account. The minimum amounts or percentages that may be allocated among Investment Options and/or through a Self-Managed Account, and the timing and frequency of re-allocations, shall be subject to such limitations and procedures as the Board may from time to time establish. In addition, investments and trading with respect to a Self-Managed Account will be subject to any rules or restrictions adopted by the third party provider that administers such accounts. The Board, in its sole discretion, may make available to Participants certain investment guidance services or investment advice services through a third party provider. Such services may be utilized only at the voluntary election of a Participant and shall not limit a Participant's responsibility for the allocation of his or her Accounts in and among the Investment Options. The Board may eliminate the availability of such services at any time.

#### **3.6 Beneficiary Designation:**

(a) Upon enrollment, the Participant shall designate a Beneficiary to receive distributions from the Participant's Account in the event of the Participant's death. Such designation shall be made in a form and manner prescribed by the Board. A Participant may change his or her designated Beneficiary at any time, provided that an amended Beneficiary designation shall be given effect only if it is made by the Participant in a form and manner prescribed by the Board and delivered to a Plan representative (or post-marked for delivery) prior to the Participant's death. The Board may allow a Beneficiary designation or change in Beneficiary designation to be made through use of an electronic medium to the extent permitted by applicable law. The last Beneficiary designation made by the Participant may designate any person or persons as Beneficiaries, subject to the limitations in (b) below. The Board shall apply the provisions of this Section to determine the proper persons to whom payment shall be made, and the decisions of the Board shall be final, binding, and conclusive on all interested persons for all purposes.

(b) If a Participant is married or in a registered domestic partnership and the Participant designates a Beneficiary who is not the Participant's spouse or domestic partner, such designation shall not be effective unless and until the spouse or domestic partner consents in writing in a form and manner prescribed by the Board.

(c) If a Participant designates more than one Beneficiary, but does not specify the percentage of benefits for each, then all designated Beneficiaries shall be entitled to equal shares of the benefits payable upon the Participant's death.

(d) Upon the death of a Participant, if there is no designated Beneficiary, or if no designated Beneficiary survives the Participant for a period of fifteen (15) days, or if a beneficiary designation form is not in good order, 100% of the Participant's benefits shall be paid to the Participant's survivors in the following order:

- (1) The Participant's spouse or registered domestic partner.
- (2) The Participant's natural or adopted children.
- (3) The Participant's parents.
- (4) The Participant's estate.

**3.7** Additional Deferrals: An Employer may make additional Fund investments with respect to any Employee, resulting in additional credits to the Account of such Employee. Any such additional credits shall be treated as Deferrals for all purposes of the Plan. The Employer shall notify the Board of any such additional Deferrals, and each Employee for whom such Deferrals are to be made must complete a Deferral Agreement, regardless of whether elective Deferrals are to be made by such Employee.

#### **3.8 Roth Elective Deferrals:**

(a) <u>General Application</u>. Employers may elect to allow their Employees to make Roth Elective Deferrals in accordance with this Section 3.8. This Section will apply to Deferrals beginning on the effective date that an Employer adopts this Section, but in no event before March 1, 2019. As of such date, the Plan will accept Roth Elective Deferrals made on behalf of a Participant who is an Employee of an Employer that has adopted this Section. A Participant's Roth Elective Deferrals will be allocated to a separate sub-account maintained for such Deferrals as described in subpart (b) below. Unless specifically stated otherwise, Roth Elective Deferrals will be treated as Deferrals for all purposes under the Plan.

(b) <u>Separate Accounting</u>. Contributions and withdrawals of Roth Elective Deferrals will be credited and debited to a separate Roth Elective Deferral sub-account maintained within the Participant's Account. The Plan will maintain a record of the amount of Roth Elective Deferrals in each Participant's Account. Gains, losses, and other credits or charges must be separately allocated on a reasonable and consistent basis to each Participant's Roth Elective Deferral sub-account and the remaining portion of the Participant's Account under the Plan. No contributions other than Roth Elective Deferrals, adjusted in accordance with Article 5, will be credited to a Participant's Roth Elective Deferral sub-account.

(c) <u>Rollovers and Transfers</u>.

(1) Notwithstanding Sections 6.10 and 7.3, a direct rollover or transfer of a distribution from a Roth Elective Deferral sub-account under the Plan will only be made to another Roth Elective Deferral account under an applicable retirement Plan described in Section 402A(e)(1) of the Code or to a Roth IRA described in Section 408A of the Code, and only to the extent the rollover is permitted under the rules of Section 402(c) of the Code.

(2) Notwithstanding Article 7, the Plan will accept a rollover contribution to a Roth Elective Deferral sub-account only if it is a direct rollover or transfer from another Roth Elective Deferral account under an applicable retirement Plan described in Section 402A(e)(1) of the Code and only to the extent the rollover is permitted under the rules of Section 402(c) of the Code.

(3) Eligible rollover distributions from a Participant's Roth Elective Deferral subaccount are taken into account in determining whether the total amount of the Participant's Account balances under the Plan exceeds \$1,000 for purposes of Section 6.5 of the Plan.

(d) <u>Correction of Excess Contributions</u>. In the case of a distribution of excess contributions, the Plan shall distribute pre-tax Deferrals and earnings thereon first.

(e) <u>Contributions and Distributions</u>. In the absence of an affirmative election to the contrary, all Deferrals contributed to the Plan shall be designated as pre-tax Deferrals. Loans from a Participant's Account must be made and repaid from pre-tax available sources and/or Roth Elective Deferrals, subject to any ordering rules established by the Board. Participants may

designate whether hardship withdrawals from their account will reduce pro-rata their Roth or their pre-tax available sources.

#### Article IV - Deferral Limitations

**4.1 General Limitation:** Except as provided in Section 4.2 or 4.3, a Participant's Deferrals for a Year shall not exceed the lesser of: (a) the Applicable Dollar Amount, or (b) the Participant's Includible Compensation for the Year.

The Applicable Dollar Amount is the amount established under Section 457(e)(15) of the Code as may be amended from time to time. The Applicable Dollar Amount is adjusted for the cost-of-living from time to the extent provided under Section 415(d) of the Code.

**4.2** Age 50 Catch-up Deferral Contributions: A Participant who will attain age 50 or more by the end of the Year is permitted to elect an additional amount of Deferrals as permitted under the Code ("Age 50 Catch-up"). The maximum annual Age 50 Catch-up is the amount established by the Code (as indexed) and is subject to the limitations of Section 414(v) of the Code. The maximum dollar amount of the Age 50 Catch-up Deferrals is adjusted for the cost-of-living from time to time to the extent provided under the Code.

**4.3** Special Section 457 Catch-up Limitation: If a Participant has designated a proper Normal Retirement Age in the manner established by the Board, and if the amount determined under this Section 4.3 exceeds the amount computed under Sections 4.1 and 4.2, then for one or more of the three (3) Years ending before the Year in which a Participant attains that Normal Retirement Age, the Participant's Deferrals shall not exceed the lesser of:

(a) An amount equal to 2 times the Section 4.1 Applicable Dollar Amount for such year, or

(b) The sum of:

(1) An amount equal to (A) the aggregate Section 4.1 limit for the current year plus each prior calendar year beginning after December 31, 2001 during which the Participant was an Employee under the Plan, minus (B) the aggregate amount of salary and wages that the Participant deferred under the Plan during such years, plus

(2) An amount equal to (A) the aggregate limit referred to in Section 457(b)(2) of the Code for each prior calendar year beginning after December 31, 1978 and before January 1, 2002 during which the Participant was an Employee (determined without regard to Sections 4.2 and 4.3), minus (B) the aggregate contributions to Pre-2002 Coordination Plans for such years.

4.4 **Special Rules:** For purposes of this Section 4, the following rules shall apply:

(a) <u>Participant Covered By More Than One Eligible Plan</u>. If the Participant is or has been a participant in one or more other eligible plans within the meaning of Section 457(b) of the Code, then this Plan and all such other plans shall be considered as one plan for purposes of applying the

foregoing limitations of this Section 4. For this purpose, the Employer shall take into account any other such eligible plan maintained by the Employer and shall also take into account any other such eligible plan for which the Employer receives from the Participant sufficient information concerning his or her participation in such other plan.

(b) <u>Pre-Participation Years</u>. In applying Section 4.3, a year shall be taken into account only if (i) the Participant was eligible to participate in the Plan during all or a portion of the year and (ii) salary and wages, if any, under the Plan during the year was subject to the Basic Annual Limitation described in Section 4.1 or any other plan ceiling required by Section 457(b) of the Code.

(c) <u>Pre-2002 Coordination Years</u>. For purposes of Section 4.3(b)(2)(B), "contributions to Pre-2002 Coordination Plans" means any employer contribution, salary reduction or elective contribution under any other eligible Code Section 457(b) plan, or a salary reduction or elective contribution under any Code Section 401(k) qualified cash or deferred arrangement, Code Section 402(h)(1)(B) simplified employee pension (SARSEP), Code Section 403(b) annuity contract, and Code Section 408(p) simple retirement account, or under any plan for which a deduction is allowed because of a contribution to an organization described in Section 501(c)(18) of the Code, including plans, arrangements or accounts maintained by the Employer or any employer for whom the Participant performed services. However, the contributions for any calendar year are only taken into account for purposes of Section 4.3(b)(2)(B) to the extent that the total of such contributions does not exceed the aggregate limit referred to in Section 457(b)(2) of the Code for that year.

(d) <u>Disregard Excess Deferral</u>. For purposes of Sections 4.1, 4.2 and 4.3, an individual is treated as not having deferred compensation under a plan for a prior taxable year to the extent Excess Deferrals under the plan are distributed, as described in Section 4.6. To the extent that the combined deferrals for pre-2002 years exceeded the maximum deferral limitations, the amount is treated as an Excess Deferral for those prior years.

# 4.5 Deferrals Prior To or After Severance from Employment, Including Sick, Vacation, and Back Pay:

A Participant who has not had a Severance from Employment may elect to defer accumulated sick pay, accumulated vacation pay, and back pay under the following terms and pursuant to any procedures established by the Board:

(a) Deferrals attributable to accumulated sick pay, accumulated vacation pay, and back pay may be deferred for any calendar month only if an agreement providing for the Deferral is entered into before the amounts would otherwise be paid or made available and the Participant is an Employee on the date the amounts would otherwise be paid or made available (provided, however, that prior to January 1, 2023, such Deferrals may be deferred for any calendar month only if an agreement providing for the Deferral is entered into before the beginning of the month in which the amounts would otherwise be paid or made available and the Participant is an Employee on the date the amounts would otherwise be paid or made available), or (b) Deferrals attributable to accumulated sick pay, accumulated vacation pay, and back pay may be made for former Employees with respect to compensation described in Section 1.415(c)-2(e)(3)(i) of the Income Tax Regulations (relating to certain compensation paid by the later of  $2\frac{1}{2}$ months following Severance From Employment or the end of the limitation year that includes the date of severance from employment), compensation described in Section 1.415(c)-2(g)(4) of the Income Tax Regulations (relating to compensation paid to participants who are permanently and totally disabled), and compensation relating to qualified military service under Section 414(u) of the Code.

**4.6 Correction of Excess Deferrals:** If the Deferrals on behalf of a Participant for any calendar year exceed the limitations described above, or the Deferrals on behalf of a Participant for any calendar year exceed the limitations described above when combined with other amounts deferred by the Participant under another Eligible Deferred Compensation Plan for which the Participant provides information that is accepted by the Employer, then the Deferral, to the extent in excess of the applicable limitation (adjusted for any income or loss in value, if any, allocable thereto), shall be distributed to the Participant.

4.7 Protection of Persons Who Serve in a Uniformed Service: An Employee whose employment is interrupted by qualified military service under Section 414(u) of the Code or who is on a leave of absence for qualified military service under Section 414(u) of the Code may elect to make additional Deferrals upon resumption of employment with the Employer equal to the maximum Deferrals that the Employee could have elected during that period if the Employee's employment with the Employer had continued (at the same level of salary or wages) without the interruption or leave, reduced by the Deferrals, if any, actually made for the Employee during the period of the interruption or leave. This right applies for five years following the resumption of employment (or, if sooner, for a period equal to three times the period of the interruption or leave). Effective January 1, 2007, if a Participant dies while performing qualified military service (as defined in Section 414(u) of the Code), the Participant's Beneficiaries shall be entitled to receive any additional benefits (other than additional credits described in Section 3.7 or similar benefit accruals relating to the period of qualified military service) provided under the Plan, as if the Participant had resumed employment and subsequently experienced a Severance from Employment on account of death.

#### Article V - Participant Accounts

**5.1 Crediting of Accounts:** All Deferrals and Transfers with respect to a Participant shall be credited to the Participant's Account as of the date such amounts are invested in the Fund in accordance with the procedures established by the Board. The Employer shall remit to CalPERS all Deferrals and Transfers directed by Participants to be invested in the Fund as soon as reasonably practicable after such amounts are withheld from the Participant's salary or wages or are available from the transferor plan, as applicable, and in no event longer than is reasonable for the proper administration of Participant accounts.

**5.2** Account Balances: The value, or balance, of each Participant's Account shall equal the aggregate value of the Fund investments held with respect to the Participant, based on the Investment Options selected by the Participant, and the method of valuation established by the Board. Each Participant shall periodically receive a statement which shows his or her Account balance and summarizes any credits or other transactions since the preceding statement. In the event that an individual has participated in this Plan by reason of employment with two or more Employers, separate Accounts shall be maintained for such individual with respect to each employment relationship.

#### **Article VI - Distributions**

6.1 Commencement of Distributions upon Severance from Employment or Attainment of the Applicable RMD Age without a Severance from Employment: Upon a Participant's Severance from Employment with an Employer (including a Severance from Employment described in Section 6.8), or upon a Participant's attainment of the Applicable RMD Age without a Severance from Employment with an Employer, the Participant may elect to receive distributions under one of the optional distribution forms described in Section 6.2, or the Participant may elect a deferred commencement date under Section 6.3. If no election is made by the date the Participant attains the Applicable RMD Age (or, if later, 30 days after the Participant's Severance from Employment), distributions shall commence in accordance with Section 6.9. If a Participant who has not attained the Applicable RMD Age becomes a Rehired Employee, then following the date of rehire, he or she shall no longer be entitled to commence distributions on account of a Severance from Employment occurring prior to becoming a Rehired Employee. Further, any outstanding distributions which commenced prior to such a Participant becoming a Rehired Employee shall be discontinued following the date of such Participant's rehire to the extent such distributions were on account of a Severance from Employment.

**6.2 Optional Distribution Forms:** Prior to the commencement date under Section 6.1 or Section 6.3, as applicable, the Participant may elect to have his or her Account distributed in one of the following forms:

(a) a single lump sum payment;

(b) installment payments for a period of years (payable on an annual, semi-annual, quarterly, or monthly basis) which extends no longer than the life expectancy of the Participant or such longer period as permitted under Section 6.9(b);

(c) partial lump sum payment of a designated amount;

(d) any combination of the above forms of distribution; or

(e) such other form of installment payments as may be approved by the Board consistent with the limitations of Section 6.9.

The Participant or a Beneficiary, after a Participant's death, may request a change to the form of

distribution initially elected under this Section. Any such election must be made in writing at least 60 days prior to the effective date of the change. The Participant's distributions made under a subsequent election shall be under one of the optional distribution forms described in this Section.

**6.3** Deferred Commencement Date: A Participant may elect a deferred commencement date for part or all of the Participant's Account balance at any time prior to the calendar year the Participant attains the Applicable RMD Age (or, if later, 30 days after the Participant's Severance from Employment). Such date may not be later than the Participant's Required Beginning Date.

**6.4 In-Service Distributions From Rollover Account:** If a Participant has a separate account attributable to rollover contributions to the Plan, the Participant may at any time elect to receive a distribution of all or any portion of the amount held in the rollover account under any optional distribution form described in Section 6.2.

**6.5 Cash-outs and Combined Payments:** The Board reserves the right to adopt guidelines, which shall be uniformly applied to all Plan Participants and Beneficiaries, under which Account balances below a specified level may be distributed in a lump sum upon Severance from Employment or at a deferred commencement date and installment payments below a specified amount may be combined and paid at less frequent intervals (but not less frequently than annually).

Subject to the limitations of Sections 457(e)(9)(A) and 457(d)(3) of the Code and Section § 1.457-6(e)(1) of the Treasury regulations, and at the direction of the Board, a Participant's Account balance shall be distributed in a lump sum without the Participant's consent, provided that (i) the Participant's Account balance does not exceed \$1,000, (ii) no Deferral has been credited to the Participant's Account in the preceding twenty-four (24) months, and (iii) no prior payment has been made to the Participant under Section 6.1 or this Section 6.5.

#### 6.6 Unforeseeable Emergency Distributions:

In the event of an Unforeseeable Emergency prior to or after the commencement of distributions, a Participant may apply to receive that part of the value of the Participant's Account that is reasonably needed to satisfy the emergency need, including any income tax resulting from the distribution. Payment will not be made to the extent that the financial hardship may be satisfied through cessation of Deferrals, insurance or other reimbursement, or a liquidation of other assets to the extent such liquidation would not itself cause severe financial hardship. The Board, or its delegee, shall determine whether a Participant is entitled to a distribution on account of an Unforeseeable Emergency, provided that the Board, or its delegee, may require the Employer to substantiate an Unforeseeable Emergency at its request.

**6.7 Death Benefits:** Upon the Participant's death, the Participant's remaining Account balance shall be payable to the Beneficiary as soon as reasonably practicable after the Participant's death, unless the Beneficiary elects a deferred commencement date that is consistent with the limitations set forth below. Prior to the Beneficiary's commencement date, the Beneficiary may elect to receive

the Participant's remaining Account balance under any optional distribution form described in Section 6.2, provided that the elected distribution form satisfies the limitations set forth below.

(a) If the Participant dies prior to the Participant's Required Beginning Date and the Beneficiary is the Participant's surviving spouse, the commencement date shall be no later than the last day of the Year in which the Participant would have attained the Applicable RMD Age (or, if later, the Year immediately following the Year of the Participant's death) and shall be paid over a period that does not exceed the Beneficiary's life expectancy using the single life table in Section 1.401(a)(9)-9, A-1, of the Income Tax Regulations for the spouse's age on the spouse's birthday for that year.

(b) If the Participant dies prior to the Participant's Required Beginning Date and the Beneficiary is not the Participant's surviving spouse, the entire Account balance shall be distributed no later than (i) the last day of the Year which includes the fifth (5th) anniversary of the Participant's death, or (ii) if distributions to the Beneficiary commence by the last day of the Year immediately following the Year of the Participant's death, the entire Account balance shall be distributed over the Beneficiary's life expectancy determined in the year following the year of the Participant's death using the single life table in Section 1.401(a)(9)-9, A-1, of the Income Tax Regulations for the Beneficiary's age on the Beneficiary's birthday for that year, reduced by one for each year that has elapsed after that year.

(c) If the Participant dies after the Participant's Required Beginning Date or after the commencement of distributions in the form of an annuity, the Beneficiary may not elect to defer the commencement of death benefits, and the Participant's remaining Account shall be distributed at least as rapidly as under the method selected by the Participant.

(d) In the event that a Beneficiary dies before all payments to the Beneficiary have been completed, the remaining value of the Participant's Account shall be paid to the estate of the Beneficiary in a lump sum.

(e) If there are two or more Beneficiaries, the provisions of this Section 6.7 and Section 6.9 shall be applied separately to each Beneficiary with respect to that Beneficiary's share in the Participant's Account.

(f) No Beneficiary shall have any right of recovery against the Employer or the Plan for any distributions that are made in the name of the Participant before a Plan representative has been duly notified of the Participant's death.

(g) If the Participant dies on or after January 1, 2022, the following additional limitations shall apply, in addition to the limitations set forth in Sections 6.7(a)-(e):

(1) Unless the designated Beneficiary meets the requirements of an eligible designated beneficiary, as defined in Code Section 401(a)(9)(E), and whether the Participant dies before or after distribution has begun, a Participant's entire interest will be distributed to the designated Beneficiary by December 31 of the calendar year containing the tenth anniversary of the Participant's death.

(2) An eligible designated beneficiary, as defined in Code Section 401(a)(9)(E), may receive distributions over the life of such eligible designated beneficiary, subject to any other limitations set forth in Sections 6.7(a)-(e).

(3) This Section 6 shall be subject to all applicable requirements specified under Code Section 401(a)(9)(H), Section 401 of the Setting Every Community Up for Retirement Enhancement Act of 2019 and any regulatory guidance issued thereunder.

**6.8** Uniformed Service Member Distributions: Effective January 1, 2009, a Participant who is performing service in the uniformed services (as defined in chapter 43 of title 38, United States Code) and on active duty for a period of more than 30 days will be treated as if the Participant has experienced a "Severance from Employment" under Section 6.1 above, in accordance with Section 414(u)(12)(B) of the Code. Such Participant may elect to receive a distribution of all or a portion of the Participant's Account; provided, however, that the Participant will not be required to take a distribution under this Section 6.8. A Participant who elects to receive a distribution pursuant to this Section 6.8 will be precluded from making Deferrals to the Plan during the six (6) month period beginning on the date of such distribution.

#### 6.9 Latest Distribution Date:

(a) In general. In no event shall any distribution under this Section 6 begin later than the Required Beginning Date. If distributions commence in the calendar year following the later of the calendar year in which the Participant attains the Applicable RMD Age or the calendar year in which the Severance from Employment occurs, the distribution on the date that distribution commences must be at least equal to the minimum installment payment for the year that the Participant has a Severance from Employment (determined under paragraph (b)) and an amount at least equal to the minimum installment payment for the year after Severance from Employment (determined under paragraph (b)) must also be paid before the end of the calendar year of commencement.

(b) Minimum installment amount. Unless a lower amount is permitted under Code Section 401(a)(9), the minimum installment amount is the amount payable equal to a fraction of the Account balance equal to one divided by the distribution period set forth in the Uniform Lifetime Table at Section 1.401(a)(9)-9, A-2, of the Income Tax Regulations for the Participant's age on the Participant's birthday for that year. If the Participant's age is less than age 70, the distribution period is 27.4 plus the number of years that the Participant's age is less than age 70. The Account balance for this calculation (other than the final installment payment) is the Account balance as of the end of the year prior to the year for which the distribution is being calculated.

(c) Extended 2020 RMDs. Notwithstanding Section 6.7 and this Section 6.9, a Participant or Beneficiary who would have been required to receive a RMD in 2020 (or paid in 2021 for the 2020 calendar year for a Participant with a required beginning date of April 1, 2021) but for the enactment of Section 401(a)(9)(I) of the Code ("2020 RMDs"), and who would have satisfied that requirement by receiving distributions that are either (1) equal to the 2020 RMDs, or (2) one or

more payments (that include the 2020 RMDs) in a series of substantially equal periodic payments made at least annually and expected to last for the life (or life expectancy) of the Participant, the joint lives (or joint life expectancies) of the Participant and the Participant's designated Beneficiary, or for a period of at least 10 years ("Extended 2020 RMDs"), will not receive those distributions for 2020 unless the Participant or Beneficiary affirmatively elects, at the time and in the manner prescribed by the Board, to receive such distributions. In addition, notwithstanding other provisions of the Plan, and solely for applying the direct rollover provisions of the Plan, 2020 RMDs and Extended 2020 RMDs are treated as eligible rollover distributions for 2020.

#### 6.10 Distributions/Rollovers from Fund:

(a) In general. Except as otherwise provided herein, all distributions shall be made directly from the Fund to the Participant or Beneficiary. To the extent required by law, income and other taxes shall be withheld from each benefit payment, and payments shall be reported to the appropriate governmental agency or agencies.

(b) Rollover Distributions. A Participant, the surviving spouse of a Participant (or a Participant's former spouse who is the alternate payee under a domestic relations order, as defined in Section 414(p) of the Code), or a nonspouse Beneficiary who is entitled to an eligible rollover distribution may elect, at the time and in the manner prescribed by the Board, to have all or any portion of the distribution paid directly to an eligible retirement plan specified by the Participant in a direct rollover.

(c) Definitions. For purposes of this Section 6.10, an eligible rollover distribution means any distribution of all or any portion of a Participant's Account balance, except that an eligible rollover distribution does not include (a) any installment payment under Section 6.2 for a period of 10 years or more (b) any distribution made under Section 6.6 as a result of an unforeseeable emergency, or (c) for any other distribution, the portion, if any, of the distribution that is a required minimum distribution under Section 401(a)(9). In addition, an eligible retirement plan means an individual retirement account described in Section 408(a) of the Code, an individual retirement annuity described in Section 408(b) of the Code, a qualified trust described in Section 401(a) of the Code, an annuity plan described in Section 403(a) or 403(b) of the Code, an eligible governmental plan described in Section 4057(b) of the Code, or a Roth individual retirement account described in Section 408A of the Code, that accepts the eligible rollover distribution, provided however that with respect to a nonspouse Beneficiary who is entitled to an eligible rollover distribution, an eligible retirement plan shall mean an individual retirement account or an individual retirement annuity that will be treated as an inherited IRA.

#### 6.11 Qualified Birth or Adoption Distributions:

(a) A Participant's Account may be distributed on or after January 1, 2020 as a qualified birth or adoption distribution. A qualified birth or adoption distribution is any distribution of up to \$5,000 (or such other amount as may be announced by the Internal Revenue Service in generally applicable guidance) from the Plan to a Participant if made during the one-year period beginning

on the date the child of the Participant is born or the legal adoption by the Participant of an eligible adoptee is finalized. A distribution of up to \$5,000 (or such other amount as may be announced by the Internal Revenue Service in generally applicable guidance) for each child can be made with respect to multiple births and adoptions if the distribution is made within the one-year period following the date on which the children are born, or the adoptions are finalized. An eligible adoptee is defined as any individual who either: (1) has not attained age 18, or (2) is physically or mentally incapable of self-support. An individual is physically or mentally incapable of self-support if they are unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment that can be expected to result in death or to be of long continued and indefinite duration, as supported by medical evidence. Absent actual knowledge to the contrary, the Board may accept the Participant's certification that any requested distribution under this Section satisfies the above-described criteria.

(b) A Participant who receives one or more qualified birth or adoption distributions under the Plan is entitled to recontribute the distributions (not to exceed the amount of the distributions) if the Participant is eligible to make a rollover contribution to the Plan at the time of recontribution. A Participant who makes a recontribution to the Plan will be treated as having received the distributions in an eligible rollover distribution and as having transferred the amount to the Plan in a direct trustee-to-trustee transfer within 60 days of the distribution.

#### 6.12 Temporary Coronavirus-Related Distributions:

(a) A Participant's Account may be distributed as a coronavirus-related distribution. A coronavirus-related distribution is any distribution made from the Plan on or after January 1, 2020, and before December 31, 2020, to a qualified individual, as defined in Section 2202(a)(4)(A)(ii) of the Coronavirus Aid, Relief, and Economic Security Act ("CARES Act") and Section 1B of Notice 2020-50, which does not exceed, in the aggregate, \$100,000.

(b) A Participant who receives a coronavirus-related distribution that is eligible for tax-free rollover treatment may recontribute all or a portion of the coronavirus-related distribution at any time during the three-year period beginning the day after the date of a coronavirus-related distribution made under the Plan. The recontribution of a coronavirus-related distribution that is eligible for tax-free rollover treatment and made within the three-year period described above will be treated as a rollover contribution to the Plan.

#### Article VII – Transfers and In-Plan Conversions

**7.1** Acceptance of Transfers: Participants who are participants in another Eligible Deferred Compensation Plan may transfer assets to the Plan as provided in this Section 7.1. Such Transfer is permitted only if the other plan provides for the direct transfer of each Participant's interest therein to the Plan, and shall be in cash or other property that the Board accepts for investment in the Fund. The Transfer shall be credited to the Participant's Account Balance and shall be held, accounted for, administered and otherwise treated in the same manner as a Deferral by the Participant under the Plan, except that the Transfer shall not be considered a Deferral in determining the maximum Deferral amounts under Article IV.

#### 7.2 Eligible Rollover Contributions to the Plan:

(a) A Participant who is an Employee and who is entitled to receive an eligible rollover distribution from another eligible retirement plan may request to have all or a portion of the eligible rollover distribution paid to the Plan. The Board may require such documentation from the distributing plan as it deems necessary to effectuate the rollover in accordance with Section 402 of the Code and to confirm that such plan is an eligible retirement plan within the meaning of Section 402(c)(8)(B) of the Code.

(b) For purposes of this Section 7.2, an eligible rollover distribution means any distribution of all or any portion of a Participant's benefit under another eligible retirement plan, except that an eligible rollover distribution does not include (1) any installment payment for a period of 10 years or more, (2) any distribution made as a result of an unforeseeable emergency or other distribution which is made upon hardship of the employee, or (3) for any other distribution, the portion, if any, of the distribution that is a required minimum distribution under Section 401(a)(9) of the Code. In addition, an eligible retirement plan means an individual retirement account described in Section 408(a) of the Code, an individual retirement annuity described in Section 408(b) of the Code, a qualified trust described in Section 401(a) of the Code, an annuity plan described in Section 403(a) or 403(b) of the Code, or an eligible governmental plan described in Section 457(b) of the Code, that accepts the eligible rollover distribution.

(c) The Plan shall establish and maintain for the Participant a separate account for any eligible rollover distribution paid to the Plan from any eligible retirement plan that is not an eligible governmental plan under Section 457(b) of the Code. In addition, the Plan shall establish and maintain for the Participant a separate account for any eligible rollover distribution paid to the Plan from any eligible retirement plan under Section 457(b) of the Code.

#### 7.3 Transfers to Other Plans:

(a) At the election of a Participant, all or a portion of the Participant's Account balance may be transferred to another Eligible Deferred Compensation Plan maintained by the Employer or another employer of the Participant. A transfer is permitted under this Section 7.3 for a Participant only under the conditions described in paragraphs (b)(2), (3) and (4) of Section 1.457-10(b) of the Income Tax Regulations (relating to post-severance plan-to-plan transfers, plan-to-plan transfers of all plan assets, and plan-to-plan transfers among eligible governmental plans of the same employer) and provided that the Participant is an employee of the entity that maintains the other Eligible Deferred Compensation Plan. Further, a transfer is permitted under this Section 7.3 only if the other Eligible Deferred Compensation Plan provides for the acceptance of plan-to-plan transfers with respect to the Participants and Beneficiaries and for each Participant and Beneficiary to have an amount deferred under the other plan immediately after the transfer at least equal to the

amount transferred, and provided that the transfer is otherwise in accordance with Section 457(e)(10) of the Code and Section 1.457-10(b) of the Income Tax Regulations.

(b) In connection with an Employer withdrawal from participation in the Plan, such Employer may direct the transfer of all Participant Account balances to another Eligible Deferred Compensation Plan, provided that such other plan accepts transfers and the transfer is otherwise in accordance with Section 457(e)(10) of the Code and Section 1.457-10(b) of the Income Tax Regulations.

**7.4 Transfer Conditions:** The Board reserves the right to limit the terms and conditions under which Transfers will be accepted from or made to other Eligible Deferred Compensation Plans, including requiring such documentation from the other plan as it deems necessary to effectuate the transfer in accordance with Section 457(e)(10) of the Code and Section 1.457-10(b) of the Income Tax Regulations and to confirm that the other plan is an Eligible Deferred Compensation Plan.

**7.5 Permissive Service Credit Transfers:** If a Participant is also a participant in a taxqualified defined benefit governmental plan (as defined in Section 414(d) of the Code) that provides for the acceptance of plan-to-plan transfers with respect to the Participant, then the Participant may elect to have any portion of the Participant's Account balance transferred to such defined benefit governmental plan as may be permitted by such defined benefit governmental plan. A transfer under this Section 7.5 may be made before the Participant has had a Severance from Employment; provided, however, that such a transfer may be made only if the transfer is either for the purchase of permissive service credit (as defined in Section 415(n)(3)(A) of the Code) under the receiving defined benefit governmental plan or a repayment to which Section 415 of the Code does not apply by reason of Section 415(k)(3) of the Code.

7.6 In-Plan Roth Conversions: The Board may, but shall not be required to, permit in-plan Roth conversions in accordance with this Section 7.6. If the Board permits in-plan Roth conversions, then a Participant, a surviving spouse Beneficiary, or a Participant's former spouse who is the alternate payee under a domestic relations order, as defined in Section 414(p) of the Code may elect to irrevocably transfer any portion of the Participant's Account (other than the Roth Elective Deferral sub-account) to the Participant's Roth Elective Deferral sub-account, in accordance with Code Section 402A(c)(4) and applicable guidance thereunder. Any such transfer shall be direct, unless the transferred amount is otherwise distributable. The Plan will maintain such records as are necessary for the proper reporting of in-plan Roth conversions. Within the Roth Elective Deferral sub-account, the Plan will separately account for amounts converted pursuant to this Section 7.6 that were distributable at the time of conversion, amounts converted pursuant to this Section 7.6 that were not distributable at the time of conversion, and rollovers from other plans of Roth elective deferrals or Roth in-plan conversion amounts, as well as any gains or losses attributable thereto. Such amounts shall be subject to the rules that apply to Roth Elective Deferrals under Section 3.8, except that the converted amount shall not be considered in determining the maximum Deferral amounts under Article IV. Provided further, to the extent such amounts were distributable at the time of conversion, they shall be subject to the distribution rules that apply to eligible rollover distributions that are paid to the Plan, and to the extent such amounts

were not distributable at the time of conversion, they shall continue to be subject to any distribution constraints that applied to the amounts prior to conversion.

#### Article VIII – Loans

**8.1 Loans:** A Participant who is an Employee of an Employer that has adopted the Loan Provision for its Employees may apply for and receive a loan from his or her Account Balance as provided in this Section 8. Any such loan may not be for an amount less than \$1,000.

**8.2** Maximum Loan Amount: No loan to a Participant hereunder may exceed the lesser of:

(a) \$50,000, reduced by the greater of (i) the outstanding balance on any loan from the Plan to the Participant on the date the loan is made or (ii) the highest outstanding balance on loans from the Plan to the Participant during the one-year period ending on the day before the date the loan is approved by the Plan (not taking into account any payments made during such one-year period), or

(b) one half of the value of the Participant's vested Account Balance (as of the Valuation Date immediately preceding the date on which such loan is approved by the Plan).

For purposes of this Section 8.2, any loan from any other plan maintained by a participating Employer shall be treated as if it were a loan made from the Plan, and the Participant's vested interest under any such other plan shall be considered a vested interest under this Plan; provided, however, that the provisions of this paragraph shall not be applied so as to allow the amount of a loan under this Section 8.2 to exceed the amount that would otherwise be permitted in the absence of this paragraph.

**8.3** Terms of Loan: The terms of the loan shall:

(a) require level amortization with payments not less frequently than quarterly throughout the repayment period, except that alternative arrangements for repayment may apply in the event that the borrower is on an bona fide unpaid leave of absence for a period not to exceed one year for leaves other than a qualified military leave within the meaning of Section 414(u) of the Code or for the duration of a leave which is due to qualified military service;

(b) require that the loan be repaid within five years; and

(c) provide for interest at a rate equal to one percentage point above the prime rate as published in the Wall Street Journal on the first business day of the month in which the loan is approved by the Plan.

#### 8.4 Security for Loan; Default:

(a) Security. Any loan to a Participant under the Plan shall be secured by the pledge of the portion of the Participant's interest in the Plan invested in such loan.

(b) Default. In the event that a Participant fails to make a loan payment under this Section 8 within the last day of the quarter following the quarter when the payment is due, a default on the loan shall occur. The rules that apply in the event of such a default shall be governed by separate loan procedures to be adopted by the Board.

8.5 **Repayment:** The Participant shall be required, as a condition to receiving a loan, to enter into an irrevocable agreement authorizing the Employer to make payroll deductions from his or her Compensation as long as the Participant is an Employee and to transfer such payroll deduction amounts to the Trustee in payment of such loan plus interest. Repayments of a loan shall be made by payroll deduction of equal amounts (comprised of both principal and interest) from each paycheck, with the first such deduction to be made as soon as practicable after the loan funds are disbursed; provided however, that a Participant may prepay the entire outstanding balance of his loan at any time (but may not make a partial prepayment); and provided, further, that if any payroll deductions cannot be made in full because a Participant is on an unpaid leave of absence or the Participant's paycheck is insufficient for any other reason, the Participant shall either repay the loan in full in accordance with procedures established by the Board, or to the extent permitted by the Board, pay directly to the Plan the full amount that would have been deducted from the Participant's paycheck, with such payment to be made by the last business day of the calendar month in which the amount would have been deducted. If a Participant with an outstanding loan balance has a Severance from Employment, the Participant may repay the loan in full in accordance with procedures established by the Board or continue to make repayments on the loan according to the loan's amortization schedule by paying such amounts directly to the Plan by the last business day of the calendar month for which the payment is attributable.

#### 8.6 Temporary Expanded Loan Provisions:

(a) The Plan permits coronavirus-related loans. A loan to a Participant may be treated as a coronavirus-related loan, subject to the requirements of this Section 8.6(a). A coronavirus-related loan is a loan made from the Plan on or after March 27, 2020, and before September 23, 2020, to a qualified individual, as defined in Section 2202(a)(4)(A)(ii) of the CARES Act and Section 1B of Notice 2020-50, which does not exceed the lesser of: (1) \$100,000 (minus outstanding Plan loans of the individual), or (2) the individual's vested benefit under the Plan and other retirement plans maintained by the Employer. Absent actual knowledge to the contrary, the Board may accept the Participant or Beneficiary's certification that the above-described criteria are satisfied.

(b) If a loan is outstanding on or after March 27, 2020, and any repayment on the loan is due from March 27, 2020, to December 31, 2020, the due date for that repayment may be delayed under the Plan for up to one year. Any payments after the suspension period must be adjusted to reflect the delay and any interest accruing during the delay, and the period of delay must be disregarded in determining the five-year period and the term of the loan under Code Sections 72(p)(2)(B) and (C).

#### **Article IX - Participant Rights**

**9.1 Participants' Interest in the Fund:** The Fund shall constitute a trust held for the exclusive benefit of Participants and Beneficiaries under the Plan. The Board is the Trustee of the Fund. No part of the corpus or income of the Fund shall be used for, or diverted to, purposes other than for the exclusive benefit of Participants and their Beneficiaries, and no Employer or creditor of an Employer shall have any interest in or claim against any part of the assets of the Fund.

**9.2** Benefits Based on Account Balances: The benefits payable to each Participant (and his or her Beneficiary) shall be measured by and limited to the amounts properly credited to the Participant's Account. A Participant shall have no claim under the Plan for any loss or diminution of his or her Account balance that is attributable to any loss in the value of the investment portfolios of the Fund that correspond with the Investment Options selected by the Participant or in connection with any other investment selected by the Participant through a Self-Managed Account.

**9.3 Nonassignability:** Except as provided in Section 9.4, and except for a return of Employer contributions made under a mistake of fact and returned without interest within twelve (12) months of the contribution, the rights of a Participant or Beneficiary under this Plan may not be sold, assigned, pledged, committed, transferred, or otherwise conveyed, and any attempt to assign or transfer rights or benefits under this Plan shall not be recognized. Except as otherwise required by law, the rights of a Participant or Beneficiary under this Plan shall not be subject to attachment, garnishment, or execution, or to transfer by operation of law in the event of bankruptcy or insolvency of the Participant or Beneficiary or otherwise.

**9.4 Transfers under Domestic Relations Orders:** To the extent required under a final judgment, decree, or order (including approval of a property settlement agreement) made pursuant to a state domestic relations law, any portion of a Participant's Account may be paid or set aside for payment to a spouse, former spouse, or child of the Participant. Where necessary to carry out the terms of such an order, a separate account shall be established with respect to the spouse, former spouse, or child who shall be entitled to choose Investment Options in the same manner as the Participant. Any amount so set aside for a spouse, former spouse, or child shall be paid out in a lump sum, unless the judgment, decree, or order directs a different form of payment. The Board also may adopt administrative procedures to permit payment to be made without regard to whether the Participant is eligible for a distribution under the Plan. Nothing in this Section 9.4 shall be construed to authorize any amount to be distributed under the Plan at a time or in a form that is not permitted under Section 457 of the Code.

**9.5 Release from Liability to Participant:** A Participant's right to receive benefits under the Plan shall be reduced to the extent that any portion of a Participant's Account has been paid or set aside for payment to a spouse, former spouse, or child pursuant to Section 9.4 or to the extent that the Employer or the Plan is otherwise subject to a binding judgment, decree, or order for the attachment, garnishment, or execution of any portion of the Participant's Account or of any distributions therefrom. The Participant shall be deemed to have released the Employer and the Plan from any claim with respect to such amounts in any case in which (i) the Employer, the Plan, or

any Plan representative has been served with legal process or otherwise joined in a proceeding relating to such amounts, (ii) the Participant has been notified of the pendency of such proceeding in the manner prescribed by the law of the jurisdiction in which the proceeding is pending for service of process or by mail from the Employer or a Plan representative to the Participant's last known mailing address, and (iii) the Participant fails to obtain an order of the court in the proceeding relieving the Employer and the Plan from the obligation to comply with the judgment, decree, or order.

**9.6 Participation in Legal Proceedings:** Neither the Employer nor any Plan representative shall be obligated to incur any cost to defend against or set aside any judgment, decree, or order relating to the division, attachment, garnishment, or execution of the Participant's Account or of any distribution therefrom. Notwithstanding the foregoing, if the Employer, the Plan, or a Plan representative is joined in any such proceeding, a Plan representative shall take such steps as it deems necessary and appropriate to protect the terms of the Plan.

#### Article X - Terms of Fund Investments

**10.1 Use of Fund:** Except as otherwise provided herein, the Fund shall serve as the exclusive investment vehicle for amounts held under this Plan. By executing an Adoption Agreement, each participating Employer shall agree to accept the terms and conditions of Fund investments set forth herein and to invest all Deferrals and Transfers with respect to its Employees in the Fund. Except as otherwise authorized by the Board, Fund investments shall be restricted to participating Employers that have adopted this Plan.

**10.2** Administration of Fund: As provided in Section 21677 of the Government Code, the Board has the exclusive control of the administration and investment of the Fund. As provided in Section 21676 of the Government Code, the Board may retain a bank or trust company to serve as a custodian for safekeeping, recordkeeping, delivery, securities valuation, investment performance reporting, or other services in connection with the investment of the Fund. In addition, the Board may retain one or more investment managers or investment advisors to manage or participate in the management of the investment portfolios of the Fund. All expenses and fees incurred in the administration of the Fund shall be treated as Plan expenses under Section 11.4.

**10.3 Investment Options:** The Board shall establish such Investment Options as it deems necessary to provide Participants with a diversified range of alternatives, including but not limited to Investment Options of the type described in Section 21673 of the Government Code. The Board may designate a "default" Investment Option under the Plan in accordance with the procedures and investment policy established by the Board. Each Investment Option shall be based upon the investment performance of one or a combination of separate investment portfolios maintained under the Fund. The Board shall specify the investment objectives and characteristics of each Investment Option and the corresponding investment portfolio or portfolios and shall provide Employers and eligible Employees with a written description of each available Investment Option. The Board, in its sole discretion, may add, eliminate, or consolidate Investment Options and corresponding investment portfolios from time to time. In the event that an Investment Option is

eliminated, the Board shall provide prior notice of such elimination, and if the Participants whose Accounts were wholly or partially allocated to that Investment Option do not make a re-allocation, the Board shall re- allocate such amounts to the available Investment Option or Investment Options that the Board in its sole discretion deems most comparable to the eliminated Investment Option.

Fund Investments: Subject to the limitations of applicable law and such further limitations 10.4 as the Board may establish, each investment portfolio of the Fund may hold any form of investment that is consistent with its investment objectives. Without limiting the generality of the foregoing, the investment portfolios may hold equity or debt securities (other than securities issued by any Employer), fixed or variable annuity contracts (including deposit administration contracts) issued by life insurance companies, certificates of deposit or fixed rate investment contracts issued by a bank or similar institution, and such short-term instruments or deposits as the Board deems necessary to satisfy the liquidity needs of the Fund. In addition, each investment portfolio may hold shares, units, or participating interests in regulated investment companies, common or collective trust funds maintained by banks or similar institutions, investment partnerships, or other pooled investment funds or trusts that may issue participating interests to Eligible Deferred Compensation Plans. Notwithstanding the foregoing, the Board, in its sole discretion, may also permit Participants to invest in a Self-Managed Account. Amounts invested by a Participant in a Self-Managed Account will be retained within the Plan and a Participant may be required to sell or exchange such investments and/or transfer amounts invested through a Self-Managed Account into one or more Investment Options made available under the Fund before taking a distribution from the Plan. Unlike the Investment Options made available under the Fund, the investments available under the Self-Managed Account are not selected, reviewed, or monitored by the Board. The Board may eliminate the availability of Self-Managed Accounts at any time.

**10.5** Valuation and Accounting: Each investment portfolio of the Fund shall be valued at least monthly, and the value of each Participant's Account shall be determined by reference to the portion of the Participant's Account allocable to each investment portfolio. The valuation of each investment portfolio shall reflect income received and accrued, realized and unrealized gains and losses, and allocable Fund expenses. The value of each Participant's interest in an investment portfolio may be measured in units, shares, or dollars. In addition, the Board shall maintain records showing the value of the Fund investments allocable to all Participants (and deceased Participants) whose entitlement to benefits under the Plan is attributable to employment with each participating Employer.

**10.6** Redemption Restrictions: No Employer shall have any right to redeem, revoke, sell, or otherwise liquidate any contribution to or investment in the Fund, except as may be necessary to:

(a) effectuate a Participant's election to transfer all or a portion of his or her Account balance to another Eligible Deferred Compensation Plan pursuant to Section 7.3(a);

(b) effectuate a transfer of Participant Account balances to another Eligible Deferred Compensation Plan as part of an Employer's withdrawal from this Plan pursuant to Section 7.3(b); or (c) correct an investment in the Fund made by reason of a mistake of fact.

Nothing in paragraphs (a) through (c), above, shall give any Employer the right to redeem, revoke, sell, or otherwise liquidate any Fund investment, unless the Board or its designee has been provided with adequate evidence of the Employer's right to do so.

#### Article XI - Administration of Plan

**11.1 Duties of Board:** Except as provided in Section 11.3, the administration of the Plan shall be under the exclusive control of the Board. The decisions of the Board shall be final, binding, and conclusive on all interested persons for all purposes. No member of the Board shall be entitled to act on or decide any matters relating solely to himself or herself or any of his or her rights or benefits under the Plan. To the maximum extent permitted by law, each member of the Board shall be held harmless for all acts performed in good faith in connection with the Plan.

**11.2 Delegation of Authority:** The Board may delegate to any individual member of the Board, any employee or employees of CalPERS, or any independent contractor the authority to act as the Board's agent with respect to any matter within the control of the Board, provided that any such delegation of authority shall be subject to revocation by the Board. Any act that the Board is required or authorized to perform under the terms of this Plan, including any communication to be made or received by the Board and the adoption of any supplementary guidelines or procedures, may be performed by an agent of the Board, provided such person is acting within the scope of that person's delegation of authority from the Board. To the maximum extent permitted by law, each employee of CalPERS shall be held harmless for any act performed in good faith in connection with the Plan.

**11.3 Duties of Employer:** In accordance with procedures established by the Board, the Employer shall be responsible:

(a) to assure that participation in the Plan is limited to Employees of the Employer and to make the Plan available to all eligible Employees;

(b) to assure that Deferrals are properly deducted from the salaries and wages of participating Employees and remitted on a timely basis and as soon as reasonably practicable to the Fund, in no event longer than is reasonable for the proper administration of Participant accounts, and to report the amount of such Deferrals on Employee's wage statements in the manner required under applicable law;

(c) to assure that Deferrals, taking account of amounts deferred under any other Eligible Deferred Compensation Plan maintained by the Employer, do not exceed the limitations described in Article 4;

(d) to approve distribution elections and applications, except as otherwise provided herein, in accordance with the requirements of Article 6;

(e) to assure that an Employee's Severance from Employment described in Section 1.24 is reported to CalPERS timely and in the manner prescribed by the Board.

(f) to provide the Board with such information and in such form as the Board deems necessary for the proper administration of the Plan; and

(g) to carry out such other responsibilities as the Employer and the Board may agree.

**11.4 Plan Expenses:** The expenses of administering the Plan and Fund, including (i) expenses incurred by the Board in the administration of the Plan and Fund, (ii) fees and expenses approved by the Board for investment advisory, custodial, recordkeeping, and other plan administration and communication services, and (iii) any other expenses or charges allocable to the Plan or the Fund that have been approved by the Board shall be charged to the Fund or, as appropriate, to a particular Investment Option or Investment Options under the Fund and shall be reflected in Participants' Account balances as provided in Section 5.2. Brokerage fees, transfer taxes, and any other costs incident to the purchase or sale by the Fund of securities or other investments shall be deemed to be part of the cost of such securities or investments or deducted in computing the sales proceeds therefrom and shall be accounted for accordingly. Notwithstanding the foregoing, the Board reserves the right, as provided in Section 21675 of the Government Code, to enter into arrangements with Employers under which specified administration costs are borne by such Employers or charged against additional Deferrals under Section 3.7 at the time invested in the Fund.

**11.5** Communications from Participants: All enrollments, elections, designations, applications and other communications by or from an Employee, Participant, Beneficiary, or legal representative of any such person regarding that person's rights under the Plan shall be made in the form and manner established by the Board and shall be deemed to have been made and delivered only upon actual receipt by the person designated by the Board to receive such communication. Neither the Board nor the Employer shall be required to give effect to any such communication that is not made on the prescribed form and in the prescribed manner and that does not contain all information called for on the prescribed form. The Employer shall promptly furnish the Board or its designee a copy of any such communication that is delivered or transmitted to the Employer.

**11.6** Communications to Employers: All notices, statements, reports, and other communications from the Board to any Employer shall be deemed to have been duly given when delivered, including electronic delivery, or when mailed by first class mail, to the official of the Employer who has been designated by the Employer in connection with its Adoption Agreement (or as a modification of the information provided in connection with its Adoption Agreement) to receive such communications.

**11.7** Communications to Participants: All notices, statements, reports, and other communications from the Board or an Employer to any Employee, Participant, Beneficiary, or legal representative of any such person shall be deemed to have been duly given when delivered,

including electronic delivery, or when mailed by first class mail, to such person at his or her last mailing address appearing on the Plan records.

**11.8 Time Periods:** As necessary or desirable to facilitate the proper administration of the Plan and consistent with the requirements of Section 457 of the Code, the Board may further restrict the time periods during which a Participant or Beneficiary is required to make any election under the Plan, including the making or amending of a Deferral Agreement, the making or amending of Investment Option selections, the election of distribution commencement dates or distribution forms.

#### Article XII - General Provisions

**12.1 Amendment:** Subject to the requirements of the Government Code, the Board reserves the right at any time to amend or modify the Plan without the consent of any Employer, Participant, or Beneficiary. The Board shall give notice of any such amendment or modification to participating Employers. Except as may be required to maintain the status of the Plan as an Eligible Deferred Compensation Plan under Section 457 of the Code or to comply with other applicable law, no amendment or modification shall impair any individual's right to benefits under the Plan or expand any Employer's obligation to provide benefits with respect to amounts previously credited to Participants' Accounts.

**12.2** Effect on Employment: Nothing contained herein shall give any Employee the right to be retained in the employment of an Employer or affect the right of an Employer to terminate any Employee's employment.

**12.3 Binding Contract:** The terms of this Plan, as duly amended from time to time, shall constitute a contract between each Participant and the Employer and shall be binding, as applicable, upon their heirs, administrators, trustees, successors, assigns, and Beneficiaries.

**12.4 Supplementary Information and Procedures:** Any explanatory brochures, pamphlets, or notices distributed by the Board to Employees, Participants, Beneficiaries, or Employers shall be distributed for information purposes and shall not override any provision of this Plan or give any person any claim or right not provided for under this Plan. Notwithstanding the foregoing, to the extent that the terms of this Plan document authorize the Board to adopt supplementary guidelines or procedures, any publication announcing such guidelines or procedures may be relied upon by the persons to whom it is distributed, unless and until modified by a subsequent publication. Any procedural requirement described in any such publication shall be binding upon the Employee, Participant, Beneficiary, or Employer, as applicable, to the same extent as if such requirement were set forth in this Plan document.

**12.5 Incompetence of Payee:** If an Employer or the Board shall find that any person to whom any amount is payable under the Plan is unable to care for his or her affairs, is a minor, or has died, any payment due him or her, or his or her estate, may be paid to his or her spouse, a child, a relative, or any other person maintaining or having custody of such person, unless a prior claim

therefor has been made by a duly appointed legal representative. Any such payment shall be a complete discharge of all liability under the Plan thereof.

**12.6 Inability to Locate Beneficiary:** Each Participant and each Beneficiary designated by the Participant (to the extent the Participant is deceased) shall be responsible for filing with the Board the Participant's or Beneficiary's (as applicable) current mailing address and other contact information that may be requested by the Board. Any communication, statement or notice addressed to a Participant or Beneficiary at the mailing address on file with the Board shall be binding on the Participant and each Beneficiary for all purposes of the Plan. If the Board notifies a Participant or Beneficiary at such mailing address that the Participant or Beneficiary is entitled to payment, and the Participant or Beneficiary fails to claim his or her benefits or is not able to be located after reasonable diligence within three (3) years after such notification, the Participant's or Beneficiary's benefits shall escheat to the state of California, to the extent permitted by applicable law.

**12.7** Applicable Law: This Plan shall be construed under the laws of the State of California and in conformity with the requirements of Section 457 of the Code and all regulations thereunder applicable to Eligible Deferred Compensation Plans.

#### RESOLUTION TO AMEND AND RESTATE THE CALIFORNIA PUBLIC EMPLOYEES' DEFERRED COMPENSATION PLAN

WHEREAS, the Board of Administration ("Board") of the California Public Employees' Retirement System ("CalPERS") administers the California Public Employees' Deferred Compensation Plan, as amended and restated effective March 1, 2019 (the "Plan");

•WHEREAS, section 12.1 of the Plan reserves to the Board the right to amend or modify the Plan at any time;

WHEREAS, the Chief, Pension Contract & Prefunding Programs Division ("PCPP"), holds delegated authority to conduct and approve plan amendments to the Plan on behalf of CalPERS;

WHEREAS, certain changes in the law resulting from the 2019 Setting Every Community Up for Retirement Enhancement Act ("SECURE Act"), the Coronavirus Aid, Relief, and Economic Security Act ("CARES Act"), and the SECURE 2.0 Act of 2022 ("SECURE 2.0 Act") affect the Plan;

WHEREAS, the Plan was amended effective January 1, 2020 to increase the required beginning date in accordance with changes made by the SECURE Act to Internal Revenue Code section 401(a)(9) that took effect on January 1, 2020; and

WHEREAS, the Chief, PCPP, has determined that it is appropriate and necessary to amend and restate the Plan in order to incorporate certain additional changes made by the SECURE Act, the CARES Act and the SECURE 2.0 Act.

**NOW THEREFORE, BE IT RESOLVED**, that the Plan is amended and restated in substantially the form attached hereto effective as of January 1, 2020.

Executed this day of May, 2023.

CALIFORNIA PUBLIC EMPLOYEES' RETIREMENT SYSTEM BOARD OF ADMINISTRATION

By: Melody Benavides Title: Division Chief Date: 51723

#### FIRST AMENDMENT TO THE CALIFORNIA PUBLIC EMPLOYEES' DEFERRED COMPENSATION PLAN

as amended and restated effective January 1, 2020

WHEREAS, the Board of Administration ("Board") of the California Public Employees' Retirement System ("CalPERS") administers the California Public Employees' Deferred Compensation Plan, as amended and restated effective January 1, 2020 (the "Plan");

WHEREAS, section 12.1 of the Plan reserves to the Board the right to amend or modify the Plan at any time;

WHEREAS, the Chief, Pension Contract & Prefunding Programs Division ("PCPP"), holds delegated authority to conduct and approve plan amendments to the Plan on behalf of CalPERS;

WHEREAS, the Plan was amended and restated effective January 1, 2020 in order to incorporate certain changes made by the 2019 Setting Every Community Up for Retirement Enhancement Act, the Coronavirus Aid, Relief, and Economic Security Act, and the SECURE 2.0 Act of 2022 ("SECURE 2.0 Act");

WHEREAS, the Plan currently provides that participating employers may elect to allow their employees to make Roth Elective Deferrals (as defined in the Plan) to the Plan; and

WHEREAS, the Chief, PCPP, has determined that the Plan should be amended to permit the Plan to accept Roth Elective Deferrals from all participants, regardless of whether their employees previously elected to allow their employees to make Roth Elective Deferrals, in order to incorporate certain additional changes made by the SECURE 2.0 Act.

NOW THEREFORE, the Plan is hereby amended as follows, effective January 1, 2024:

1. Section 3.8(a) is amended to read as follows:

"(a) <u>General Application</u>. Effective January 1, 2024, Participants may elect to make Roth Elective Deferrals in accordance with this Section 3.8. As of such date, the Plan will accept Roth Elective Deferrals made on behalf of a Participant who is an Employee of an Employer. A Participant's Roth Elective Deferrals will be allocated to a separate subaccount maintained for such Deferrals as described in subpart (b) below. Unless specifically stated otherwise, Roth Elective Deferrals will be treated as Deferrals for all purposes under the Plan."

2. In order to set forth the provisions of the Plan in a single document, changes made by this amendment may be incorporated into the most recent restatement of the Plan.

Executed this day of December, 2023.

**CALIFORNIA PUBLIC EMPLOYEES' RETIREMENT SYSTEM BOARD OF ADMINISTRATION** 

By: Melody Benavides Title: Division Chief Date: 12

# Your choice, simplified.

For more information on how to put the CalPERS 457 Plan to work for your employees, please contact CalPERS at **800-696-3907**, or visit the <u>Employer Resource Center at calpers-sip.com</u> today!





**RESOLUTION NO.** <u>2025-01</u>

#### A RESOLUTION OF THE BOARD OF TRUSTEES OF THE DELTA MOSQUITO AND VECTOR CONTROL DISTRICT

#### A RESOLUTION APPROVING ADOPTION OF CALPERS 457 PLAN

**WHEREAS**, Delta Mosquito and Vector Control District desires to establish an additional deferred compensation plan for the benefit of its employees; and

**WHEREAS**, the Board of Administration (the Board) of the California Public Employees' Retirement System (CalPERS) has established the CalPERS 457 Plan (the "CalPERS 457 Plan") which may be adopted by a governmental employer the employees of which are public employees; and

**WHEREAS**, Delta Mosquito and Vector Control District believes that the CalPERS 457 Plan and the investment options available thereunder will provide valuable benefits to its employees; and

**WHEREAS**, the Board has appointed Voya Financial® (the Plan Recordkeeper) to perform recordkeeping and administrative services under the CalPERS 457 Plan and to act as the Board's agent in all matters relating to the administration of the CalPERS 457 Plan;

**NOW, THEREFORE, BE IT RESOLVED**, that Delta Mosquito and Vector Control District adopts the CalPERS 457 Plan for the benefit of its employees and authorizes and directs the General Manager to execute the attached adoption agreement on behalf of Delta Mosquito and Vector Control District and to provide CalPERS or any successor agent duly appointed by the Board with such information and cooperation as may be needed on an ongoing basis in the administration of the CalPERS 457 Plan. A copy of this resolution, the agreement, and any attachments thereto shall be on file in the office of Delta Mosquito and Vector Control District.

**PASSED and ADOPTED** by the Board of Trustees of the Delta Mosquito and Vector Control District, State of California on July 10, 2024, by the following vote:

AYES:

NOES:

ABSTAINED:

ABSENT:

Board President

Secretary

# 9. Rescind or Modify Policy 2139, District Equipment Loan Policy



July Action Items: Background Information for the Board:

# Item 9, Rescind or Modify Policy 2139, District Equipment Loan Policy

#### Introduction

The current Equipment Loan Policy of the Delta Mosquito and Vector Control District allows for the borrowing of District equipment by staff members. While this policy includes measures for accountability and oversight, it raises significant legal concerns under California law, which strictly prohibits the private use of public resources. To ensure compliance and maintain public trust, it is essential to revise or remove the policy to align with the relevant legal requirements.

#### Legal Framework

#### 1. California Constitution Article XVI, Section 6:

• This section expressly prohibits the use of public resources for private purposes. It mandates that public funds and resources be used exclusively for public benefit.

#### 2. California Government Code Section 8314:

 This section explicitly prohibits state and local officials and employees from using public resources for personal purposes. It defines "public resources" broadly, including equipment, and specifies that such use is a misuse of public resources, punishable by fines and penalties.

#### 3. California Penal Code Section 424:

 This section deals with embezzlement and the unlawful use of public funds, including public property, for personal gain. It underscores the seriousness of unauthorized use, highlighting potential criminal charges.

#### **Issues with the Current Policy**

#### 1. Potential for Misuse:

 The current policy, which permits the borrowing of District equipment for potentially nonofficial purposes, could be interpreted as allowing private use of public resources. This conflicts with the constitutional and statutory prohibitions mentioned above.

#### 2. Legal and Financial Risks:

Allowing private use of public resources exposes the District to significant legal risks, including fines, penalties, and potential criminal charges against employees. It also poses financial risks if public resources are damaged or misused.

#### 3. Public Trust and Transparency:

• The appearance of private use of public resources can undermine public trust in the District's governance and management. Transparency and strict adherence to legal standards are vital to maintaining the confidence of the community we serve.

**Recommended Action:** Absent a legitimate reason for employees to borrow district equipment for public use purposes, there is no reason to maintain this policy and many reasons to rescind it. If a policy is needed to cover the use of District equipment after hours for District use, a different approach is recommended. I recommend rescinding Policy # 2139 and ceasing any lending of District equipment, if occurring, immediately.

Sincerely,

Conlin Reis, General Manager

# Delta Mosquito and Vector Control District

# POLICY TITLE DISTRICT EQUIPMENT LOAN POLICY

POLICY NUMBER 2139

2139.1 This policy is created to provide the staff borrower with a clear understanding of his/her responsibilities when borrowing District property. This policy is a privilege and may be halted at any time by the General Manager due to:

- Negligence
- Misuse
- Unreported Damage
- Failure to Return Items in a Timely Manner
- Failure to appropriately request the equipment
- Excessive wear and tear

2139.2 The borrower shall assume all responsibility for any damage to borrowed equipment and is responsible for informing the District Manager of the damage at the earliest opportunity. The borrower also assumes all liability for any injury which may occur from the use of this equipment while in his/her custody.

2139.3 Some equipment is vital to District operations and therefore may only be available for a limited amount of time, or not at all.

2139.4 The return date must be decided and agreed upon before the loaning of equipment is allowed. The loan of equipment must be approved by the General Manager, direct supervisor of the employee, and supervisor in charge of the specific item of equipment.

# Delta Mosquito and Vector Control District

### POLICY TITLE DISTRICT EQUIPMENT LOAN POLICY

POLICY NUMBER 2139

# District Equipment Loan Request Form

Item requested:	
Department Item belongs to:	
Date of Loan Start:	
Date of Loan End:	

I agree to the following items:

Initials	
	To return the equipment item no later than the time listed as Loan End above.
	To assume all responsibility for any damage to the equipment.
	To report any damage to the District at the earliest opportunity.
	To reimburse up to the full cost of the equipment depending on the amount of damage/loss of equipment.
	To assume all liability or injury which may occur from the use of this equipment while in my custody.

	Date:
Request made by: (Print name)	
Request made by: (signature)	
Direct Supervisor (signature)	
Equipment Supervisor (Signature)	
General Manager (signature)	

### 10. Manager's District Review and First Steps Presentation

# **11. Board of Trustees Member Comments**

# 12. Future Agenda Items

# 13. Adjournment